National Council for the Professional Development of Nursing and Midwifery

An Evaluation of the Extent and Nature of Nurse-Led/Midwife-Led Services in Ireland

April 2005



National Council for the Professional Development of Nursing and Midwifery

An Chomhairle Naisiúnta d'Fhorbairt Ghairmiúil an Altranais agus an Chnáimhseachais

Mission Statement of the National Council

The Council exists to promote and develop the professional role of nurses and midwives in order to ensure the delivery of quality nursing and midwifery care to patients/clients in a changing healthcare environment.

© National Council for the Professional Development of Nursing and Midwifery 2005

Published by National Council for the Professional Development of Nursing and Midwifery 6-7 Manor Street Business Park Manor Street Dublin 7

> t: 353 1 882 5300 f: 353 1 868 0366 e: admin@ncnm.ie w: www.ncnm.ie

National Council for the Professional Development of Nursing and Midwifery

An Evaluation of the Extent and Nature of Nurse-Led/Midwife-Led Services in Ireland

April 2005



National Council for the Professional Development of Nursing and Midwifery

An Chomhairle Náisiúnta d'Fhorbairt Ghairmiúil an Altranais agus an Chnáimhseachais

Contents

Foreword	5
Introduction	7
Terms of reference	7
Definition of nurse-led care	7
Definition of midwife-led care	7
Methodology	7
Chapter One - Literature Review	9
Irish policy documents	9
Literature review - Nurse-led services	12
Literature review - Midwife-led services	15
Chapter Two - Findings from the questionnaire	17
Methodology	17
Questionnaire design	17
Discussion	24
Chapter Three - Findings from the focus groups	25
Evolution of nurse-led/midwife-led services	25
Factors critical to the establishment of nurse-led/midwife-led services	26
Barriers to establishing nurse-led/midwife-led services	26
The benefits to the service and/or the patient/client of nurse-led/midwife-led services	27
Discussion	27
Chapter Four	
Conclusions	29
Framework for establishing nurse-led/midwife-led services	31
References	35
Appendices	41

Foreword

The National Council for the Professional Development of Nursing and Midwifery is pleased to publish this report which has evaluated the extent and nature of nurse-led/midwife-led services in Ireland.

As part of the National Council's ongoing contribution to the professional development of nursing and midwifery in Ireland, this report highlights the increasingly significant role nurses and midwives are playing in the Irish health care system. The diversity and multiplicity of the services would suggest that nurses and midwives are able to respond to patient/client need in a flexible and appropriate manner, allowing the development to occur within a multidisciplinary context. To date these initiatives have been driven by service need and a desire by nurses and midwives, both at senior and clinical levels to improve the quality of the patient/client care.

The findings from this report identify areas where nurse-led/midwife-led services have been developed and emerged in response to an identified gap in service. This service need was identified in different ways, but perhaps the most important method of identification was asking the patient/client what type of service they wanted and in what manner did they want it to be delivered. As the Health Service Reform Programme continues to be implemented new ways of working will be identified.

The National Council would like to sincerely thank all those nurses and midwives who took the time to fill out the questionnaire or attended the focus groups. Without their help and expertise this report would have been impossible.

In addition, my thanks are due to my colleagues, Kathleen McLellan, Head of Professional Development, Christine Hughes, Jenny Hogan who led this research, Mary Farrelly and Georgina Farren, Professional Development Officers.

Yvonne O'Shea CHIEF EXECUTIVE OFFICER

Introduction

The introduction of a health service reform programme within the Irish healthcare system combined with changing epidemiological and demographic patterns are providing challenges to the traditional delivery of health care. To this end the National Council has undertaken a study to identify the extent and nature of nurse-led/midwife-led services in Ireland. This will provide a baseline to review the potential of such services to support healthcare delivery in Ireland. International experience indicates that such services can add value to the system meeting patient/client and population health need.

It is important to stress that nurse-led/midwife-led care is delivered within the context of a multidisciplinary/partnership approach. No one healthcare professional should or indeed could provide all the care that patients/clients may need over the span of a lifetime. All healthcare professionals work within the scope of their own competency, having the ability to recognise boundaries and knowing when to consult with other professions.

This research identifies the drivers and barriers to date within the Irish healthcare system for nurse/midwife-led care. Utilisation of the research results, the literature and international experience provides the template for a business plan approach which can be utilised by service to develop nurse/midwife-led care in appropriate settings.

The terms of reference for the study were as follows:

- To examine the literature pertaining to nurse-led/midwife-led care services
- To identify the extent of nurse-led/midwife-led care services in Ireland
- To make recommendations on future areas for developments for nurse-led/midwife-led care in Ireland

There are many differing definitions of nurse-led/midwife-led care and for the purposes of this study the following definitions as outlined by the National Council will be used:

Definition of nurse-led care

Nurse-led care is distinct from nurse-coordinated or nurse-managed services. Nurse-led care is provided by nurses responsible for case management, which includes comprehensive patient/client assessment, developing, implementing and managing a plan of care, clinical leadership and decision to admit or discharge. Patients/clients will be referred to nurse-led services by nurses, midwives or other healthcare professionals, in accordance with collaboratively agreed protocols. Such care requires enhanced skills and knowledge and the nurse will need preparation in both the clinical and management aspects of the role. Such nurses will be practising at an advanced level and may be working in approved specialist or advanced practice roles¹ (National Council 2003).

Definition of midwife-led care

Midwife-led care has been defined by the Cochrane protocol as "the context of care where 'the midwife is the lead professional in the planning, organisation and control of the care given to a woman from initial booking to the postnatal period' (RCOG 2001). Within these models, midwives are, in partnership with the woman, the lead professional with responsibility for assessment of her needs, planning her care, referral to other health professionals as appropriate, and for ensuring provision of maternity services" (Hatem et al 2004).

Methodology

The methodology employed for this study consisted of focus groups, an extensive questionnaire and a literature review. The focus groups were facilitated and scribed by members of the executive of the National Council and were analysed using a content analysis approach. The questionnaire was sent to all directors of nursing and midwifery in Ireland. The questionnaire

¹ An advanced level of practice indicates that the nurse has taken on an expanded role within his/her scope of practice, inclusive of any additional training and education required.

was based on the National Council's definition of nurse-led/midwife-led care. The data from the questionnaire was double entered into an SPSS database and analysed accordingly. A random sample of practice nurses were contacted to ascertain the extent of practice nurse-led services that they offered. Finally a national and international literature review was conducted and the relevant Irish policy documents were consulted for reference to nurse-led/midwife-led services.

The report is structured as follows:

Chapter one provides an overview of the literature from a national and international perspective and from the relevant Irish policy documents that support the introduction of nurse-led/midwife-led services.

Chapter two presents the findings from the questionnaires. Respondents to the questionnaire who indicated that they had nurse-led/midwife-led services within their organisation were invited to participate in the focus groups.

Chapter three presents the findings from the focus groups.

Chapter four sets out the conclusions and a framework for establishing nurse-led/midwife-led services.

The development of nurse-led/midwife-led services in Ireland has been influenced by three factors: health policy, the changing nature of the practice environment and the need to develop services that meet patient/client needs at a local level. The aim of this section is to give an overview of the relevant contemporary policy (and others) documents that make reference to the concept of nurse-led/midwife-led care in Ireland and to examine the international nursing and midwifery literature pertaining to nurse-led/midwife-led care or services.

Irish policy documents

Report of The Commission on Nursing: A Blueprint for the Future (Government of Ireland 1998)

The Commission on Nursing made recommendations for the development of nursing and midwifery in Ireland, many of which have now been implemented. The Commission envisaged that its recommendations would facilitate the development of the role of the nurse and the midwife and the science (and art) of nursing and midwifery in a rapidly changing healthcare environment. Whilst the Commission admitted that its recommendations would not be a panacea for all ills, with a large degree of foresight, it envisaged that structural changes would take place which would enable the profession to grow and that changes in education would influence shifting role boundaries (pp 50-51). These structural changes have manifested themselves in the Health Service Reform Programme (HSRP) announced by the Minister for Health and Children in 2003. The impact of the HSRP on nurse-led/midwife-led services will be detailed below.

Quality and Fairness - A Health System For You (DoHC 2001)

Quality and Fairness (Irish health strategy) forms the context within which nursing and midwifery development will occur. Nurse-led/midwife-led services underpin the four guiding principles of the health strategy (equity, people-centredness, quality and accountability) as nurses and midwives will be adapting to the diverse and changing health needs of society generally and of the individuals within it. Nurse-led/midwife-led services will empower patients/clients to be active participants in decisions relating to their health. Nurse-led/midwife-led services will also be offered to patients/clients at the right time, in the right place and by the right healthcare professional.

The Health Strategy also identified that the health service will have the right people, with the right competencies, in the right numbers, organised and managed in the right way, to deliver the goals and objectives of the Health Strategy (p.121). The Health Strategy refers to care delivery improvements which could emerge as nurse-led/midwife-led services, and in some cases these have already begun to.

'Minor injury units will be established to ensure appropriate treatment and management of non-urgent cases and advanced Nurse Practitioners (ANPs) (Emergency) will be appointed in acute hospitals. ANPs diagnose and treat certain groups of patients independently within agreed protocols' (p.106).

Primary Care: A New Direction (DoHC 2001)

The Primary Care Strategy did not specifically explore the role of nurse-led/midwife-led services in the primary care setting, nevertheless nurse-led telephone advice and triage with appropriate decision and support systems were identified as integral to a twenty-four hour primary care service.

Review of Nurses and Midwives in the Prescribing and Administration of Medicinal Products Project

The Review of Nurses and Midwives in the Prescribing and Administration of Medicinal Products Project is a joint initiative currently being managed by An Bord Altranais and the National Council. The Steering Committee will make recommendations based on the results and analysis of the research conducted as part of the focus groups, the needs assessment survey, the exploration of need survey with stakeholders and the pilot study for collaborative prescribing. The Review and the Steering Committee's recommendations are set to further the agenda and guide the way for greater discussion and subsequent action by the government and health service providers in the introduction of nurse/midwife prescribing. A progress report was published in December 2004 and is available on the National Council's website www.ncnm.ie . The final report will be completed in June 2005.

Report of the National Task Force on Medical Staffing (DoHC 2003a)

The 'Hanly' Report, as this is more commonly known, identifies services that should be provided in local hospitals. These include nurse-led minor injury units for an extended day (e.g. 8am to 8pm, depending on the volume of cases). It recommends that patients should be triaged either to treatment on site for minor injuries, treatment by a general practitioner, or transferred to a major hospital as required. The report alludes to the role that the clinical nurse specialist and advanced nurse practitioner can play in providing nurse-led clinics. Stating:

'the CNS role is already well defined and is in keeping with the task force's concept of utilising the skills of health professionals to best effect' (p.93)

The task force sees considerable benefits in the planned development of new roles and Table 1 outlines the areas requiring further development identified in the Hanly Report.

Table 1: Potential nurse-led services identified in the Hanly Report (DoHC 2003a)

CLINICAL AREA	DESCRIPTION OF SERVICE
Minor injury clinics	Run by ANPs who assess, diagnose, treat in accordance with protocols, refer and discharge. They ensure that the patient receives the correct intervention from the appropriate person at the right time.
Pre-assessment clinics	ANPs have a key role in pre-assessment of patients undergoing elective day and in-patient procedures. Registered nurses also have a role in assessing patients not requiring a general anaesthetic.
Heart failure clinics	ANPs and CNSs have a key role in the management of 'at risk' patients following discharge until stabilised.

The Hanly Report makes recommendations that are in keeping with the philosophy of the Commission on Nursing. Consequently, the scope for enhancing the roles of nurses and midwives should be further explored with a view to identifying how such enhancement could be implemented. Hanly acknowledges that appropriate education, training and protocols are required to enhance roles. The opportunities for nurses and midwives to enhance their scope of practice and to be innovative and creative in the delivery of patient/client care are clear.

Report of the National Task Force on Medical Staffing: The Challenge for Nursing and Midwifery -A Discussion Paper (DoHC 2003b)

Following the publication of the Hanly Report the Nursing Policy Division in the Department of Health and Children published a response. The discussion paper outlines what it terms 'critical success factors' for the development of nursing and midwifery. These include: management of change, partnership, leadership, education and professional development, competence and clinical guidelines. The report states that there is considerable potential for nurses and midwives to further enhance the development of high quality patient-centred care and to influence positive patient care outcomes. The discussion paper identifies a range of possible developments for nursing and midwifery elicited from nurses and midwives in acute, psychiatric and midwifery settings; these include nurse-led admission, discharge protocols, nurse-led minor injury clinics, pre-assessment clinics, midwife-led antenatal clinics and midwife-led labour ward care.

An Explorative Study into the Expansion of Nursing and Midwifery Professional Roles in Response to the European Working Time Directive (MWHB 2003)

The nursing and midwifery planning and development unit in the (former) Mid-Western Health Board published the findings of a research study which explored the possible development opportunities for nursing and midwifery in light of the European Working Time Directive (EWTD). Thirty-eight focus groups were conducted and all grades of staff welcomed the opportunity to explore professional development possibilities stemming from the introduction of the EWTD. The findings indicated that:

- 1. enhancement of nursing and midwifery practice should involve a multi-disciplinary/-professional approach and involve all stakeholders
- 2. role development, with autonomy and decision-making capacity should be assured
- 3. appropriate support structures must be included.

Some key areas for development were suggested by focus group participants. These include:

- 1. nurse-led pre-assessment units
- 2. nurse-led admissions with the possibility of an out-of-hours triage service
- 3. nurse-led clinics in wound management/tissue viability, diabetes, heart failure, continence promotion, asthma, pain management and dermatology
- 4. nurse-led out-patients
- 5. nurse discharge planning
- 6. minor injuries units
- 7. nurse-led counselling, e.g. family therapy, cognitive behavioural therapy, etc.
- 8. total management by midwives of normal pregnancy and labour
- 9. intravenous cannulation, venepuncture, male catheterisation as part of role development
- 10. competency-based education and training for all role development.

Throughout the focus groups, nurses and midwives highlighted the need for more nurse-led/midwife-led services. They stressed, however, that any new developments must be supported by appropriate education and training and must be protocol-driven. The report can be downloaded from the National Council's web site at www.ncnm.ie.

The Scope of Nursing and Midwifery Practice Framework (ABA 2000)

The scope of nursing and midwifery practice in Ireland refers to the range, roles, functions, responsibilities and activities, for which a registered nurse or midwife has the education, competence and authority to perform. Furthermore, the scope of practice framework states that nurses and midwives are autonomous in the practise of nursing/midwifery. In order to practice autonomously nurses and midwives require authority to use their professional judgement in fulfiling their responsibilities. The nurse or midwife is accountable for his/her practice. This means that the nurse or midwife is accountable for decisions he/she makes in determining his/her scope of practice. This includes decisions to expand or not to expand his/her scope of practice (p. 21). The framework document also highlights that where expansion of practice has been considered most successful, guidelines, protocols or policies have been developed collaboratively with reference to legislation and research based literature where available.

The Scope of Practice Framework highlights the principles and values which underpin role development and expansion. It is a critical document around which nurses and midwives in Ireland have the facility to develop their role within an agreed framework.

Empowerment Narratives (DoHC 2003c)

Empowerment Narratives (2003) were complied and edited by the 'Meaning of Empowerment Sub-Group of the Empowerment of Nurses and Midwives Steering Groups – An Agenda for Change'. This publication demonstrated by means of exemplars how nurse and midwives as empowered people, have greater control over achieving the goals of the organisation, and how they can develop into confident and effective professionals. The Empowerment Narratives include details on nurse-led/midwife-led services which include:

- Nurse-led wound clinics
- "Bug busting"
- Nurse-led pre-operative screening clinic
- Activities nurse
- Substance abuse service
- Breast feeding hospital initiative
- Domino and hospital outreach home birth service
- Community-based nurse-led parentcraft
- Acute community day hospital and homecare service
- Home-based treatment team
- Community-based counselling initiative.

The details of each of these services, the project outline and the contact details of the individuals involved in the development are all outlined in the report. The report is available on the Department of Health and Children's web site at: www.dohc.ie

Framework for the Establishment of Advanced Nurse/Midwife Practitioner Posts, (National Council 2004a)

The National Council has developed a definition and core concepts for the role of the advanced nurse/midwife practitioner and it determines the requirements for nurses and midwives to be accredited as advanced nurse/midwife practitioners. To date a growing number of ANP posts², have been developed by services and approved by the National Council. The posts established reflect service need in the specific areas and their development has required creativity and leadership on behalf of all those involved.

Framework for the Establishment of Clinical Nurse/Midwife Specialist Posts Intermediate Pathway (National Council 2004b)

The National Council has developed a definition and core concepts for the role of the clinical nurse/midwife specialist, and it has determined the criteria for nurses and midwives to be approved as clinical nurse/midwife specialists. In January 2004 the National Council published the findings from an extensive research study entitled An Evaluation of the Effectiveness of the Clinical Nurse/Midwife Specialist. Following this publication, the National Council reviewed its CNS/CMS Intermediate Pathway document (2001) to reflect the research findings. The Evaluation of Effectiveness study outlined the critical areas for development and also described a process for the future development of the CNS/CMS role at local, regional and national level.

Summary

From reviewing the above documents it can be seen that recognition and support for nurse-led/midwife-led change and development is gradually increasing. Nurses and midwives are making an increasingly vital contribution to the health of population and to population health, the expansion of practice of all nurses and midwives and the advent of the clinical nurse/midwife specialist and the advanced nurse/midwife practitioner are testament to this. The Scope of Practice Framework provides the foundation for all nurses to advance and develop their practice.

Nurse-led care: A Review of the Literature

Introduction

A search for literature on Medline, Cinahl and the Cochrane database was conducted using the terms 'nurse-led', 'nurse led', 'nurse-led services', 'nurse led services', 'nurse-led care', 'nurse led care', 'nurse-led evaluation' and 'nurse led evaluation'. The search results revealed a huge volume of articles. In order to give some coherence to this volume, the literature was looked at from an 'area of practice' perspective, the aim being to demonstrate the extent of the growth of the concept of 'nurse-led' into health care and the issues that have arisen as a result of this growth. Table 2 outlines the extent of the areas of practice where nurse-led services have developed.

Table 2: Nurse-led services identified in the literature

•	nurse-led palliative care services	•	nurse-led cervical screening services
•	comparing nurse-led services to medical-led services	•	nurse-led counselling service
•	nurse-led out-patient services	•	nurse-led cystoscopy services
•	nurse-led fracture referral service	•	nurse-led wound assessment services
•	nurse-led osteoporosis clinic (regional)	•	patient satisfaction with nurse-led services
•	NHS direct – nurse-led telephone triage	•	nurse-led walk in centres
•	nurse-led pain management intervention	•	nurse-led diabetes care
•	nurse-led community services	•	outcomes from nurse-led services
•	nurse-led care for children with asthma	•	nurse-led rapid response services
•	nurse-led hypertension clinics	•	nurse-led in patient units
•	nurse-led cardioversion clinics	•	practice nurse-led management of acute minor illness
•	nurse-led secondary prevention clinics (coronary heart disease)	•	

² To date 15 ANPs in emergency, 3 in cardiothoracic, 1 in cardiology, sexual health, rheumatology, breast care and primary care have been accredited.

It is evident from the literature that nurses are now beginning to measure the impact of their interventions on health care in order to demonstrate the efficacy of their interventions (National Council 2004a, National Council 2004b, Horrocks et al 2002, Garvican et al 1998, Hall-Smith et al 1997, Mackintosh and Bowles 1997). Forster and Young (1996) in a systematic review of whether the nurse practitioner working in primary care can provide care equivalent to that provided by doctors, concludes that patients are at least as satisfied with care at the point of first contact with nurse practitioners as they are with that given by doctors.

Richardson and Cunliffe (2003), in an article identifying the reasons for the accelerated development of nurse-led services, suggest that changes in health policy and the modernisation of health services give opportunities to reconsider and challenge traditional health care roles and professional boundaries. The need to provide nurse-led services and expand the nursing role can be seen as innovative and creative ways of re-balancing the workforce-demand mismatch caused by the EWTD. Finally, and perhaps more importantly, Richardson and Cunliffe identify that pressures to improve quality and ensure value for money have made healthcare professionals think about their roles and about new ways of working.

The proliferation of articles and the wide range of nurse-led services are self-evident. Most of the articles are concerned with evaluating and auditing the impact of the nurse-led service. The majority of authors state that proper educational preparation, clinical support and medical 'buy in' are crucial for the successful implementation and sustainability of nurse-led services. Table 3 illustrates nurse-led studies that have examined the impact of the nurse-led intervention from an outcomes perspective. The type of outcomes and the authors are included.

OUTCOMES	AUTHORS	METHODOLOGY (M=) S	AMPLE NUMBERS (N=)
Cost effectiveness of nurse-led services	York et al (1998)	M= RCT	N=103
Reduced hospital re-admissions	Blue et al (2001)	M=RCT	N=165
	Madge et al (1998)	M=RCT	N=201
Clinical effectiveness of nurse-led intervention	Mac Lellan (2004)	M= Pre-test-post-test	N=800
	Fitzmaurice et al (2000)	M=RCT	N=224
	Denver et al (2003)	M=RCT	N=102
	McHugh et al (2001)	M=RCT	N=98
	Rando (2000)	M=Client satisfaction	N=46
	Twinn & Cheng (1999)	M=Case Study	N=50
	Mackintosh & Bowles (1997)	M=Two Stage Baseline Survey	N=100 & 106
	Williams et al (2002)	M=Literature review	
Decreased hospital admission	Campbell et al (1999)	M=RCT	N=1343
Improved health	Campbell et al (1999)	M=Two Stage Baseline Survey	N=71 & 141
	Leveille et al (1998)	M=RCT	N=201
Improved quality of life	Harrison et al (2002)	M=RCT	N=192
	Gallagher (1999)	M=Audit	N=43
	Smith et al (2002)	M=Survey	N=100
Increased length of hospital stay	Griffiths et al (2001)	M=RCT	N=175
	Wilson -Barnett et al (2001)	M=Literature Review	N=N/A
Patient perceptions	Wiles et al (2003)	M=RCT & Qualitative	N=246
Improved patient satisfaction	Dobson (1999)	M=Project management approach	N=Not given
	Sutcliffe (1999)	M=Survey	N=N/A
	Roxborough (2000)	M=Audit of Service	N=Not Given
	Pritchard & Kendrick (2001)	M=Survey	N=1900
Comparison with medical	Moore et al (2002)	M=RCT	N=203
services for improved quality	Miles et al (2003)	M=Survey	N=282
of life and patient satisfaction	Cox & Wilson (2003)	M=Literature Review	N=N/A
	Bostrom et al (2003)	M= Two-centred descriptive design	n N=46
	Delamere (2004)	M=Survey	N=200
Impact of nurse-led clinics on reducing waiting times	Currie, et al (2003) Roxborough (2000)	M=Prospective Audit	N=83

Table 3: Types of outcomes, methods and sample numbers of nurse-led evaluations

Nurse-led services in specialised areas

There is scant published evidence of nurse-led services for people with intellectual disabilities (ID); that identified was concerned with physical healthcare. It has consistently been reported that people with ID have more medical disorders than their peers (Van Schrojenstein Lanteman-de Valk et al 2000, Mencap 2004) and that they experience difficulties in accessing appropriate health services (Kerr et al 1996, National Disability Authority 2004), particularly people with ID living in the community (Marshall et al 2003a). One Irish study (Walsh et al 2000, cited by Walsh et al 2004) concluded that health checks for people with ID living in a residential setting did not take place "at the levels appropriate for this population and fell short of the frequency suggested for the general population in Ireland" (Walsh et al 2004, p289).

British authors have identified the role of learning disability nurses in nurse-led health screening and health promotion activities relating, for example, to obesity (Marshall et al 2003b) and attendance at general practitioner clinics in the community (Martin et al 2004). While it is not clear that their role and functions fully match the definition of nurse-led services stated by the National Council, it is clear that nurse-led health screening of and health promotion activities among people with ID form a vital component of an integrated, multidisciplinary and cross-sectoral approach to ensuring that this client group receives high-quality and appropriate healthcare.

Given the wide range of settings in which nurses in Ireland work with people with ID and of specialist areas in which clinical nurse specialists work (National Council 2003), nurse-led services that may develop in the future need not necessarily be restricted to the area of health screening and assessment. It is also necessary that roles relating to these services should be developed following consideration of the advantages and disadvantages of medical and social models of ID service provision, and in accordance with the principles of inclusion.

Some literature exists relating to nurse-led mental health care. Evaluative research undertaken indicates mixed findings. For example a randomised controlled trial comparing acute ward care with mental health liaison nurse-led care (including assessment, direct interventions and liaison support) for medically ill older patients in the North of England found that the intervention group were less depressed at 6-8 week follow up although overall no differences were found in general health scores (Baldwin et al 2001). The authors suggest that services which focus on particular patient groups or disorders are more likely to be effective. Nurse-led case management for patients presenting with deliberate self-harm in Accident and Emergency was compared with routine nurse management in a randomised controlled trial in the United Kingdom, and no significant reduction in the overall admission rate was found (Clarke et al 2002). A comparison of nurse-led self-harm assessment service to individuals who self-harm in Cardiff suggested no difference between the service provided by a mental health nurse and psychiatrist, concluding that, an expansion of the nursing role in this area would not lead to a decrease in the quality of the service provided (Griffin 2001).

Descriptions of nurse-led mental health services exist in areas such as liaison (NIMHE 2005), sexual health and mental health (The Northern and Yorkshire Regional NHS Modernisation Programme 2005) and urgent/crisis referrals (Tummey 2001). However, it is often difficult to elicit the extent to which services described are nurse-led as opposed to being nurse provided.

The relative lack of literature on nurse-led services in mental health care may be a reflection of the generally low level of evaluative research carried out on nursing interventions coupled with an issue of some services which are nurse-led failing to be defined as such.

Nurse-led care in older people services have emerged in the literature in a number of areas. A nurse-led fast track system introduced in an emergency department has empowered emergency nurses using new care pathways to directly refer high-risk elderly patients thus supporting a higher standard of care for this group (Davies-Grey 2003). Nurse consultants in the United Kingdom are leading falls clinics, wards and caseloads both in hospital and primary care (Clegg and Mansfield 2003).

Diverse areas of nurse-led care for children are developing. In response to the increase in the incidence in asthma, nurse-led clinics are showing a positive impact on quality of care (Gallagher 1999). A nurse-led intervention improved the management of urinary tract infections in children, was valued by doctors and parents, and may have prevented some renal scarring (Coulthard et al 2003). In the case of children receiving chemotherapy nurse-led assessment of children has resulted in fewer delays and improved care (Boyer and Whiles 2004).

An increasing emphasis in primary care over the last decade has resulted in a proliferation of nurse-led services being offered to patients/clients by practice nurses. Areas such as acute back pain, epilepsy, acute minor illnesses and diabetes management have been introduced (Pritchard and Kendrick 2001, Mills et al 2002, Breen et al 2004, Kinealy et al 2004).

Midwife-led Care: A Review of the Literature

Introduction

In the industrialised world, there are two conflicting models of maternity care. The basis of the psychosocial approach is that childbirth is 'natural' and that the majority of women can have a normal safe childbirth without medical intervention. The traditional biomedical approach is based on the idea that childbirth is only normal in retrospect and that 'normal' childbirth requires medical supervision to ensure safety for the woman and her baby. International developments in the maternity services so that the individual needs of women are more central to the provision of care have gained momentum in recent years (Winterton Report 1992, DoH 1993). These initiatives recommend a lead role and greater responsibility for midwives in the provision of maternity care and support the development of packages of midwife-led care in which midwives provide care throughout the antenatal, intrapartum and postnatal period.

In recent years in Ireland there has been a debate amongst stakeholders regarding choice for women in the models of care available in childbirth. It appears that women wish to have access to more care options regarding childbirth which include flexibility in accessing services, natural birth, midwife-led care and the development of community services (DoH 1997, DoHC 2001a). Maternity care consumer groups such as Irish Childbirth Trust, Home Birth Association, the Association for Improvements in the Maternity Services (AIMS), Midwifery Birth Alliance, La Leche League of Ireland and Baby Milk Action Ireland have raised awareness about choice and control in pregnancy and childbirth. Midwives have indicated that they are cognisant of, and willing to be responsive to, women's needs and many are now actively seeking opportunities to practise midwifery within settings that provide autonomy and alternative choices for models of maternity care (Commission on Nursing 1998, An Bord Altranais 2000).

The World Health Organisation (WHO 1999) has also acknowledged midwife-led care as 'the most appropriate and cost effective type of health care provider to be assigned to the care of women in normal pregnancy and birth, including the risk assessment and recognition of complications'.

International literature: Midwife-led care

Various models of midwife-led care have been established in order to achieve a women-centred service which would include continuity of care and the provision of increased choice and control over their care for women and their partners. These include team midwifery models where a team of midwives provide care to a group of women throughout the antenatal, intrapartum and postpartum period, 'case-load' midwifery in which midwives carry their own personal case-load and a midwife-run antenatal clinic. There are a number of articles which discuss policies, protocols, guidelines, booking and exclusion criteria for midwife-led care (Campbell et al 1999, Walker et al 1995). There is also a substantial amount of audit and research examining outcomes in terms of neonatal and maternal morbidity and obstetrical intervention rates (Waldenstrom and Turnbull 1998, Campbell et al 1999, Hatem et al 2004). At a psychosocial level women who have availed of midwife-led care expressed increased satisfaction with their childbirth experience, felt better prepared for parenthood as they had better support and advice and experienced less postnatal depression (Biro et al 2003, Watts et al 2003). Not only have women expressed increased satisfaction with their maternity care, but also studies indicate that midwives' job satisfaction has increased as midwife-led care facilitates midwives to use their midwifery skills more appropriately and enables them to practice in an autonomous manner (Todd et al 1998, Saunders et al 2000).

OUTCOMES	AUTHORS	METHODOLOGY (M=) &	x SAMPLE NUMBERS (N=)
Increased maternal satisfaction	Walker et al (1995)	M= interview	N= 38
	Turnbull et al (1996)	M= RCT	N= 1299
	Hundley et al (1994)	M= RCT	N= 2844
	Shields et al (1998)	M= RCT	N= 1299
	Byrne et al (2000)	M= RCT	N= 201
	Hodnett et al (2001)	M= Cochrane Review	
	Morgan et al (1998)	M= Survey	N= 469
	Biro et al (2003)	M= RCT	N= 1000
Decreased intervention rates	Harvey et al (1996)	M= RCT	N= 194
	Waldenstrom et al (1997)	M= RCT	N= 1860
	Waldenstrom & Turnbull (1998)	M=RCT	Review of 7
	Saunders et al (2000)	M=RCT	N= Unknown
	Biro et al (2000)	M= Mixed	N= 1000
Increased breastfeeding rates	O'Connell and Cronin (2002)	Descriptive analysis	N= N/A
Increased job satisfaction	Todd et al (1998)	M= Survey	N= 92
	Saunders et al (2000)	M= Mixed	N= Unknown
Increased cost-effectiveness	Young et al (1997)	M= RCT	N= 1299
	Harvey et al (1996)	M= RCT	N= 194
	Villar et al (2004)	M= Cochrane Review	N= N/A
Increased continuity of care	Waldenstrom et al (2000)	M= RCT	N= 1000
	Hodnett (2001)	M= Cochrane Review	N= N/A
	Homer et al (2002)	M= Survey	N= 1089

Table 4: Types of outcomes, methods and sample numbers of midwife-led evaluations

Midwife-led care in Ireland

In 1994 a Report on the Mother and Infant Care Scheme recommended that a pilot domiciliary service scheme be established (DoH 1994). On foot of this the health boards set up an Expert Group on Domiciliary Births to review the current provision of maternity care and recommend developments which would facilitate more choices for women (Expert Group 1998). This resulted in the establishment of a number of pilot schemes in midwife-led care, two of which were hospital-based, the other being community-based. Of these pilot schemes The National Maternity Hospital Domino and Outreach Home Birth Service commenced in 1999 and has been evaluated very positively (Women's Health Unit 2001). An evaluation of the other pilot sites is being conducted at present.

In 2001, in response to consumer demand for a midwife-led service, an integrated hospital and community midwifery service was developed in the (former) South-Eastern Health Board, an evaluation of which is currently being undertaken. Following the recommendations of the Kinder report (Kinder 2001), two midwifery-led units were established in the (former) North-Eastern Health Board. Booking-in of women for maternity care in these units commenced in June 2004, the first baby being born in the midwife-led unit in Cavan on 30th December 2004. Following the successful implementation and evaluation of these units, further pilot sites in this region may be an option for consideration (Comhairle Na nOspideal 2003).

The Report of the National Task Force on Medical Staffing (DoHC 2003a) suggest that this is a time of significant potential for midwives. The Challenge for Nursing and Midwifery Discussion paper (DoHC 2003b) highlighted the high potential for midwife-led clinics and it was clear that midwives themselves felt that they delivered a very specific package of care. It was suggested that the establishment of more midwifery-led clinics would offer women holistic and continuous care as well as providing a "best value for money" service.

Methodology

A total of 306 questionnaires were sent to all the directors of nursing services in Ireland. Of these, 147 were returned completed giving a total response rate of 48%. Of the 20 questionnaires sent to the directors of midwifery services in Ireland, four were returned giving a response rate of 20%.

Questionnaire design

The questionnaire design was based on the definition of nurse-led/midwife-led care as outlined by the definitions provided on page 7. The aim of this was to identify if services were actually nurse-led or midwife-led as opposed to being nurse-run or midwife-run or nurse-managed or midwife-managed. The questionnaire was also designed to elicit from respondents the type of nurse-led/midwife-led service provided, if any. The questionnaire also aimed to elicit at what grade the nurse/midwife running the service worked, and to what level they were educated. A further aim was to discover from whom the nurses/midwives running the service could receive referrals and to whom they could refer. Finally the questionnaire asked respondents to indicate if they audited the service and if they were planning to introduce nurse-led/midwife-led services in the future. Full details of the questionnaire are available in Appendix 1.

Demographics

Each respondent was asked to identify the care setting in which they were working or employed. Of respondents to the questionnaire, 24% were from a community setting, 39% from a hospital setting and 27% came from both hospital and community setting. The remaining 10% of services which did not 'fit' the criteria included in this response were areas such as the blood transfusion service, the prison service, nursing homes, community nursing units, residential services for persons with intellectual disability, day centres and hospices (see Figure 1).

Figure 1: Care setting of respondents



Each organisation was then asked to identify to which hospital band they belonged: 22% indicated that they did not belong to any banding system. The band that was represented most frequently was band 4 at 21%; bands 3 and 5 were the next highest with 16% respectively, band 1 was represented at 14% and finally band 2 was represented by 12% of respondents. Figure 2 demonstrates the bands.



Figure 2: Hospital bands

Each respondent was asked to supply the number of beds in their organisation, taking into account that some services do not have 'beds' per se. Figure 3 demonstrates the majority of beds per organisation: 20% of respondents to the questionnaire did not have beds and over 25% of respondents had between 50-100 beds.





Of the respondents to the questionnaire, 47% stated that they provided nurse-led or midwife-led services, whilst 50% stated that they did not have nurse-led or midwife-led services (3% were missing values). A small number of questionnaires had accompanying letters stating that they could not fill in this question as they did not fit the criteria.

Question five asked respondents to identify their nurse-led or midwife-led services and to estimate how long each services had been established. Out of a total of 147 responses, 69 (47%) claimed to have nurse-led/midwife-led services; of these, 44 (30%) fitted the criteria for nurse or midwife-led care as indicated by the National Council (see page 7), and the remaining 25 (17%) did not fit the criteria. These criteria were determined by the definition of nurse-led care and midwife-led care given on page 7. Table 5 outlines the type of services and the number of years established.

Table 5: Type of nurse-led/midwife-led service and number of years established³

TYPE OF SERVICE	YEARS ESTD.	TYPE OF SERVICE	YEARS ESTD.
Adult tissue viability clinic	7	Generic counsellor: day services	3
Adult pre-assessment day-care	4	Haemodialysis treatment	9
Adult renal transplantation services	2	Heart failure	2-3
Adult haemocomatosis services	2	Home care	15
Adult pre-admission elective orthopaedics	2	Hospice day care	11
Admission/discharge planning	3-4	Hypertension – practice nursing	1-6
Advanced nurse practitioner - Accident & Emergency service	1-5	Immunisation – adult and paediatric - practice nursing	1-6
Alzheimer unit	3	Lactation	5
Assessment and case management in	5	Leg ulcer clinic	1-4
day hospital	3	Lymphoedema clinic	7
Asthma and COPD – practice nursing	1-6	Men's health – practice nursing	1-7
Asylum seekers	2	Midwife-led clinic	6
Behavioural therapy	10	Midwifery-led unit	6 months
Blood collection clinics	2	Midwives' clinics	15
Breast and cervical screening -		Minor injuries A & E	3
practice nursing	1-6	Occupational health	5
Bone densitometry estimation	3	Oncology services	3
Cardiac rehabilitation	5	Outreach clinics	3
Care of older person	34	Paediatric cystic fibrosis clinic	14
Case management	4	Paediatric/Adult diabetes services	14/5
Case manager home subvention	1-10	Paediatric endocrinology	9
Child health screening	20-34	Paediatric respiratory services	7
Cognitive behavioural therapy	5-8	Paediatric/adult dermatology	7/2
Colposcopy services	10	Paediatric urology	7
Community alcohol & substance misuse		Palliative care	1-14
counselling services	7-8	Parent craft classes	6-25
Community midwifery	6	Patient assessment	4
Community programme (mental health)	18	Postnatal care	15
Continence assessment and management	4-28	Primary care wound clinic	5
Continuing care	Not specified	Rehabilitation services	2
Convalescence care	3-15	Residential services	50
Cryotherapy – practice nursing	Not specified	Respiratory nurse clinic	3
Coronary heart disease – practice nursing	1-6	Respite care	3-10
Day care services	8-20	Respite care (crisis)	8
Deliberate self-harm liaison nurse	2	Respite care (planned)	8
Dementia care services	6	School screening	34
Developing care plans	4	School service	20-30
Diabetes – including practice nursing	3		
Disability service – assessment of needs	34	Substance abuse service	Not stated
Ear care- practice nursing	1-6	Therapeutic aphaeresis service Travellers' service	8 months
Early transfer home scheme	2		10
Extended care	20	Travel health – practice nursing	1-4
Family therapy services	4	Urodynamics Wound care including log ulger	3
Family planning – practice nursing	1-6	Wound care including leg ulcer management – practice nursing	1-0
Foetal assessment	4	Women's health – practice nursing	1-6

 $^{\scriptscriptstyle 3}$ Where the years are in multiples, more than one organisatiom provides this service.

Respondents to the questionnaire were asked to identify if the nurse-/midwife-led service was able to admit or discharge their patients/clients. The caveat to this question was that admit or discharge can mean admission or discharge to hospital or to another health care institution or from the episode of care that the nurse or midwife was providing (see appendix 1 for the questionnaire). Figures 4 and 5 outlines the percentages of services, which had nurse-led/midwife-led services that were able to admit or discharge. A large percentage (54%) of respondents to the questionnaire who had nurse-led/midwife-led services were not sure if patients could be admitted or discharged into or across services.









The following table indicates the health and social care professionals to whom the nurse or midwife providing the nurse-led/midwife-led service can refer.

Table 5: Health and social care workers to whom the nurse/midwife running the nurse-led/midwife-led service could refer⁴

HEALTH CARE WORKER	%
Physiotherapist	33
CNSs/CMSs	30
Occupational therapist	29
Speech therapist	23
Radiographer	7
Community mental health nurse	24
Public health nurse	36
Social worker	33
Psychologist	16
Medical staff	37
GP	43
Consultant	28
No-one	1
Other ⁵	15

Table 6 outlines the percentage of respondents to whom the nurse or midwife running the nurse-led/midwife-led service could refer.

Table 6: Health and social care professionals (including patients/clients) who could refer to the nurse-led/midwife-led clinic⁴

HEALTH CARE WORKER	%
Dietician	13
Physiotherapist	20
CNSs/CMSs	20
Occupational therapist	18
Speech therapist	14
Radiographer	4
Public health nurse	31
Community mental health nurse	22
Social worker	22
Psychologist	13
GP	40
Consultant	33
Other medical staff	26
Staff nurses/midwives	24
CNM 1, 2 or 3	22
Self-referrals by patients/clients	24
No-one	1
Other ⁵	9

⁴ The percentage totals 120%, as some respondents ticked more than one option.

⁵ 'Other' included ophthalmology, chaplaincy, transplant co-ordinators, chiropody, etc.

The following table outlines the grade of staff working in nurse-led or midwife-led services. It can be noted that there is a more or less even distribution among staff nurses (25%), clinical nurse/midwife specialists (25%) and clinical nurse/midwife managers (27%). The advanced nurse/midwife practitioner accounts for 2% of services.

Table 7: Grade of staff providing the nurse-led/midwife-led service

GRADE OF STAFF	%
CNM/CMM	27
Staff nurse/midwife	25
CNS/CMS	25
Other	20
ANP/AMP	2
None	1

Nurses and midwives who run nurse-led and midwife-led services were asked to identify the type of education or training they had received in order to prepare them for the role. The following table indicates the type and percentage of education and training that they received. Forty-three per cent stated that they had received training or education to equip them for the role, while nine per cent stated they had received none. Forty-eight per cent stated they did not know if they or their staff had received training and/or education.

Table 8: Type of training and education nurses and midwives received to equip them for the role



Respondents were asked to summarise the type of nurse-led/midwife-led services that they were intending to introduce or develop in the future. Thirty-four per cent were planning to introduce and/or develop nurse-led/midwife-led services in the future and 35% stated that they were not planning to introduce any nurse-led or midwife-led services. Table 9 is an outline of some of those responses.

Table 9: Types of service that directors of nursing/midwifery are intending to introduce in the future

• Reality and cognitive behavioural sessions with self-referral to	Nurse-led palliative care via ANP posts
a nurse-led community service	Nurse-led care and case management (community care)
Day care service for older people	Nurse-led falls clinic
Rheumatology	Nurse-led therapeutic day hospital
ANP in pain management	Midwifery-led ultrasonography service
Out-patient cataract nurse-led clinic	Midwifery-led postnatal clinics
Incontinence care in the older person	Midwifery-led breast-feeding support groups
Health promotion in the elderly	Nurse-led mental health promotion
Nurse-led pre-admission assessment service	Nurse-led mental health rehabilitation
Challenging behaviour	
• Holistic assessment of clients in the community who may	 Nurse-led cognitive behavioural psychotherapy Review of home birth services
require elderly care	
• Nurse-led clinic for clients with enduring mental health	Eating, drinking and swallowing nurse-led clinic
difficulties	Nurse-led glaucoma clinic
• Nurse-led liaison service (Mental health)	Nurse-led therapeutic apheresis
Nurse-led community detoxification programme	Nurse-led platelet apheresis
Diabetic ANP	Radiation induced toxicities (ANP)
Nurse-led colposcopy clinic	• Midwifery-led services for healthy women likely to have a
Midwife-led early miscarriage clinic	normal pregnancy and labour
Midwife-led admissions and discharges	Nurse-led child health primary screening
Tissue viability	Nurse-led cervical cytology clinics
Nurse-led pain management	Nurse-led community leg ulcer clinics
Midwifery-led clinics	Nurse-led enuresis treatment clinic
Pre-/post-HIV testing sessions	Nurse-led child health service

Of respondents to the questionnaire, who indicated that they provided nurse-led/midwife-led services, 65% (n=52) stated that they audited the service they provided while 35% (n=28) stated that they did not. Table 10 illustrates what aspects of the services were audited.

Table 10: Example of audit type

•	Numbers of patients seen by service	•	Numbers of review visits
•	Numbers of patients seen by nurse	•	Location of contact ⁶
•	Mean waiting time to be seen by nurse	•	Type of care delivered at each episode
•	Registration time to time seen by nurse	•	Diagnostic reason for referral
•	Triage time to time seen by nurse	•	Most common diagnostic group
•	Registration time to time departing the service	•	Average number of visits per month
•	Consumer satisfaction survey	•	Busiest months
•	Demographic profile of service users	•	Details of procedures carried out
•	Numbers of new referrals on a monthly basis	•	Numbers of patients discharged per month
•	Number of visits to the service	•	Re-admission rates
•	Geographic area of residence of service user	•	Length of stay
•	Referring consultant	•	Impact of nurse-led service on patient/client knowledge of
•	Type of referral		own disease
•	Reason for referral	•	Preferred time of day for intervention

Some of the audit findings were used to compile an annual report which the nurse or midwife running the service used to identify further areas for development. For example, one of the reports stressed that service provision could be improved with the development of written agreed multidisciplinary referral guidelines and discharge criteria.

Discussion

It is evident from the results of the questionnaire that there are already many developments in nurse-led/midwife-led care in Ireland. The diversity and multiplicity of the services would suggest that nurses and midwives are able to respond to patient/client need in a flexible and appropriate manner. These developments have taken place in a multidisciplinary working environment and reflect a certain level of multidisciplinary collaboration and co-operation. To date these initiatives have been driven by service need and a desire by nurses and midwives, both at senior and clinical level to improve the quality of the patient/client care. It is also evident that as these services develop they are being audited and measured for clinical effectiveness and patient satisfaction. These developments can only increase and maximise the therapeutic potential of nursing and midwifery for patients and clients. What is also evident is that nurses and midwives have been leading some of these services for many years, and some such services have only recently been established. How these services were established and what kind of barriers and benefits were accrued are now discussed in light of the findings of the focus groups.

CHAPTER THREE Findings from the Focus Groups

Three focus groups were held with nurses and midwives from the services who had indicated in the questionnaire that they would be willing to discuss issues surrounding the establishment of their nurse-led/midwife-led service. The focus groups were held in Sligo, Cork and Dublin and a total of 26 nurses and midwives participated. The profile of the attendees was mostly assistant directors of nursing and midwifery and clinical nurse/midwife managers.

Evolution of nurse-led/midwife-led services

Several themes emerged from the focus group discussions, such as factors leading to the evolution of the nurse-led/midwife-led services, barriers to their success and the benefits to patients/clients. These are now discussed.

Service need and patient/client demand

The overarching reason for the establishment of nurse-led/midwife-led services was service need. This need was identified in various ways, for example, patients/clients were asked what kind of service they wanted. This led to the establishment of evening diabetic clinics, leg ulcer clinics and evening lactation and midwife-led ante-natal clinics. Out-of-hours services that were responsive to patient/client need were successfully introduced as a direct consequence of requests from the patient/client.

'out of hours service, responsive and flexible to service need'

Changing demographics

Changing demographics, population trends and unhealthy lifestyles were cited as reasons why new services were introduced. The need for more patient/client education and general health promotion were other factors that were considered important when establishing nurse-led/midwife-led services.

'the need to educate patients, patients come in under the medical model and go home with no education, no understanding of their illness, but now nurses are educating, and have more time to talk and promote the patients own well being'

Reduction of waiting times

The need to reduce patient/client waiting times and to provide a rapid access service was identified as a priority in some areas. The need to utilise the CNS/CMS role to its maximum potential led to the establishment, in some areas, of nurse-led/midwife-led clinics, allowing the CNS/CMS to assess, plan, treat, discharge and/or refer on.

'nurses took ownership (for developing nurse-led services) themselves, went off and trained and identified the need and pushed ahead, good pioneers'

Patient/client choice

Midwife-led ante-natal and foetal assessment clinics were established when a gap in the existing services was identified, the aim being to give women a choice regarding their pregnancy pathway.

'a lot of the midwifery-led service has come from professional development, increasing confidence thanks to education and literature'

Continuity of care

Nurse-led clinics for enduring mental illness were set up because it was identified that there was poor continuity of care for clients and clients were seeing a different healthcare professional at each episode of care. Nurse-led early intervention clinics were established as early intervention in enduring mental health has become accepted as best practice.

Factors critical to the establishment of nurse-led/midwife-led services

Accurate data

The factors critical to the successful implementation of nurse-led/midwife-led service included, identification of service need from an empirical perspective. This meant that evidence needed to be collected and analysed, the service need identified, a business case written and presented in an informed manner to senior managers.

'research it properly and show the need – make a business case and sell it to the managers/establishment competing for resources'

Collaboration

Looking at international best practice and using it to support the business case was seen as essential. Once the proposal became embedded in the service plan the factors that became important included: communication, collaboration and education. Collaboration from the medical staff and senior nursing staff was seen as crucial for the proposal to succeed.

Role definition

Clarification of new roles was seen as essential for the service to succeed; where there was lack of clarity about the role, resistance to the introduction of the service increased. Education of the relevant health care personnel about the new service was deemed essential if the service was to succeed.

Desire to change

Motivation, commitment and a desire to change hospital/service culture were cited as important factors when attempting to drive change. Having a vision for how services can be improved was cited as critical to successful implementation.

'I went into the director of nursing with an idea and was given the go ahead'

'there is another way forward'

'be willing to embrace change'

Finally, having the appropriate person with the appropriate level of knowledge and the appropriate expertise to lead the service was seen as essential from a clinical perspective.

Barriers to establishing nurse-led/midwife-led services

Financial constraints

In many instances the barriers to establishing nurse-led/midwife-led services were the inverse of the factors that were deemed essential to their establishment. However, participants at the focus groups identified some other concerns about setting up this type of service. Not least of these were financial resources. Participants agreed that it was often difficult to get 'development monies'; in many instances services had to be developed out of existing resources. Finance for professional development of staff was also difficult to access. Participants suggested that nurses and midwives preparing to lead nurse-led/midwife-led services need to have the necessary and appropriate education and/or training.

Lack of accurate data

A lack of Irish research and audit data was cited as an obstacle, as were a lack of availability of baseline demographic data and a lack of access for audit training and education.

Prescribing difficulties

The lack of authority to prescribe was cited as a barrier, although participants did acknowledge that this was being addressed. Participants nevertheless suggested that the lead-in time, and the time it will take to build a cohort of nurses and midwives who can prescribe will continue to be a barrier for the foreseeable or medium term future.

Location of service

The location of the nurse-led/midwife-led service was a consideration for participants. It was considered crucial that the service location must have an accessible, reliable and effective public transport system and be close to other amenities such as a pharmacy and for easy referral to other health care professionals. A 'one stop shop' type of service was considered the ideal scenario for the patient/client journey to be as seamless as possible.

Referral mechanisms

Poor mechanisms for referring patients/clients to and from the nurse-led/midwife-led service was cited as a barrier. Referral pathways and clear and unambiguous criteria for the reason for referral needed to be established and agreed by all the health care professionals involved in the service.

Risk taking

Lack of a 'risk taking' attitude to developments and a leaning towards medical dependency were regarded as further obstacles to developing nurse-led/midwife-led services. It was thought that a lot of nurses preferred to work 'alongside' medical staff rather than as autonomous individuals. Some of the participants felt very strongly that nurses and midwives use the scope of practice as a means to declare what they can't do rather than what they can.

Trade union factor

Participants expressed the view that the trade unions were, in some instances, concerned with financial remuneration for expanding the scope of practice rather than professional development opportunities. However, in some other instances the help of the trade unions in driving an agenda for change was seen as advantageous.

Clerical support

Finally, the lack of clerical support, especially for collecting and entering data for audit purposes, was seen as a considerable obstacle.

The benefits to the service and/or the patient/client of nurse-led/midwife-led care

The benefits to the patient/client included: increasing the continuity of care, patient's/client's being able to spend more time at home (assuming this was their preferred location), increased support in the community, relieving the burden on the acute services, improved symptom control, increased cover 'out of hours' and increased support after discharge. Patient/client compliance with medication and better education of patients/clients were seen as positive benefits.

The benefits to the service obviously overlapped with benefits to the patient/client on the basis that if the patient/client receives a satisfactory service then the service itself feels the overall benefit or effect. Direct measurable benefits to the service included better efficiency, improved patient/client journey, more individualised care, reduction in readmission rates, developing shared areas of expertise, disseminating and publishing the same. The streamlining of documentation was cited as a benefit as it reduced duplication and prevented the patient's/client's from telling their story over and over again to different health care professionals.

'people are voting with their feet, enquiries to join the service are increasing, word of mouth: people are coming back to book in'

All of the benefits to the service and to the patient/client that were identified in the focus groups were not merely anecdotal; most of the services has measured the effect the nurse-led/midwife-led care was having on the users and on the service by audit or questionnaire. Finally, job satisfaction and greater autonomy for staff were cited as associated benefits from the service.

Discussion

It is clear that nurse-led and midwife-led services currently available are wide-ranging, flexible and patient/clientcentred. However, taking into account the recommendations from the Hanly report and the international literature, there is more scope for further development of these services to develop in Ireland. Nurse-led/midwife-led care could and should be developed in areas such as emergency, primary care, care of the older person, pre-assessment clinics, pain management, out-patient clinics, ante-natal care, 'total midwifery' care. With the proper educational and structural supports nurse-led and midwife-led services have great potential to expand within the framework of national policy. Service need would appear to be the single most important factor when the concept of nurse-led/midwife-led care was being considered.

CHAPTER FOUR Conclusions and Framework for Establishing Nurse-led/Midwife-led Services

Conclusion

The expectations and demands of patients/clients receiving healthcare and healthcare services are shifting. Changes in the provision of healthcare are having an impact on the demands made on the healthcare providers, and have resulted in shorter stays in acute services, quicker access to hospital services and greater emphasis on the importance of providing community care. A significant factor in support of such changes is the creative use of nursing and midwifery expertise and a more effective utilisation of nurses and midwives experience and skills.

A culture of innovation, a desire to change and a motivated workforce are pre-requisites to any organisational change. If any of these factors are absent the process could be protracted and the desired outcomes may not be achieved. A proactive and strategic approach to the development of nurse-led/midwife-led services can ensure that services are developed in a flexible, innovative and creative manner to meet the evolving needs of patients/clients using the health services.

Increasingly, nurses and midwives are required to enable and co-ordinate care and liase across acute and community boundaries. New models of care are being developed that require all clinical and health and social care practitioners to challenge traditional professional boundaries. Important changes are taking place in the public's expectations and perceptions of the health services.

This research study has identified the extent and type of nurse-led/midwife-led services that are currently available in Ireland. It is evident that nurse-led/midwife-led services are continually being developed and refined by nurses and midwives who want to improve the quality of patient/client care. What is also evident is that these nurse-led/midwife-led services have evolved because of a gap in the service or because the service was not meeting the expectations of the patient/client. This can be evidenced by the introduction of evening and out of hour's clinics that suit the patient/client and not necessarily the health care professional delivering that service.

It is evident that depending on location and type of service nurse/midwife-led care can be delivered by nurses at generalist, specialist and advanced practice level. For those providing the clinical care it is critical that the clinical leadership is provided at the appropriate level. Ongoing professional development needs of the nurses and midwives delivering nurse-led/midwife-led services should be supported by the organisation.

As part of the questionnaire, each respondent was asked to indicate whether their organisation was considering developing any nurse-led/midwife-led services, and if so, to outline what they might be. Table 11 outlines their responses for future development of nurse-led/midwife-led care. This is not an exhaustive list as it reflects the views of those who responded to the questionnaire; there are many other areas that may not have been considered.

The diversity and multiplicity of the services would suggest that nurses and midwives are able to respond to patient/client need in a flexible and appropriate manner, allowing the development to occur within a multidisciplinary context. To date these initiatives have been driven by service need and a desire by nurses and midwives, both at senior and clinical level to improve the quality of the patient/client care. It is also evident that as these services develop they are being audited and measured for clinical effectiveness and patient satisfaction.

The development of nurse-led/midwife-led services in a structured collaborative manner will support the emerging health service needs. Nursing and midwifery competencies will be maximised, contributing to responsive delivery of care. Patients/clients will benefit from innovative and creative systems for delivering care.

Table 11: Areas for future development of nurse-led/midwifery-led services

NURSE-LED SERVICES IN GENERAL NURSING
minor injury services
pain assessment
care of the elderly services
out patient clinics in wound care
telephone triage
pre-assessment clinics (pre surgery)
heart failure clinics
clinics in tissue viability
clinics in continence promotion
clinics in asthma
clinics in dermatology
nurse-led discharge
palliative care services
osteoporosis clinics
community services
hypertension clinics
cardioversion clinics
secondary prevention clinics (coronary care)
cervical screening services
cystoscopy clinics
walk in centres
diabetes care
holistic/alternative therapies care
NURSE-LED SERVICES IN INTELLECTUAL DISABILITY
health screening (with specific groups, e.g., the older person with ID, people with severe and profound ID, children with complex needs)
health promotion (e.g., relating to obesity, exercise)
behaviour management
mental health assessment
advocacy for people with ID attending mainstream health and mental health services
MIDWIFE-LED SERVICES
total management of normal pregnancy
total management of normal pregnancy midwife-led domino care
midwife-led domino care
midwife-led domino care midwife-led outreach home birth service
midwife-led domino care midwife-led outreach home birth service midwife-led units
midwife-led domino care midwife-led outreach home birth service midwife-led units NURSE-LED SERVICES IN PSYCHIATRY
midwife-led domino care midwife-led outreach home birth service midwife-led units NURSE-LED SERVICES IN PSYCHIATRY counselling – family therapy
midwife-led domino care midwife-led outreach home birth service midwife-led units NURSE-LED SERVICES IN PSYCHIATRY counselling – family therapy counselling – cognitive behavioural therapy
midwife-led domino care midwife-led outreach home birth service midwife-led units NURSE-LED SERVICES IN PSYCHIATRY counselling – family therapy counselling – cognitive behavioural therapy addiction service
midwife-led domino care midwife-led outreach home birth service midwife-led units NURSE-LED SERVICES IN PSYCHIATRY counselling – family therapy counselling – cognitive behavioural therapy addiction service rehabilitation service
midwife-led domino care midwife-led outreach home birth service midwife-led units NURSE-LED SERVICES IN PSYCHIATRY counselling – family therapy counselling – cognitive behavioural therapy addiction service rehabilitation service attention deficit hyperactivity disorder service
midwife-led domino care midwife-led outreach home birth service midwife-led units NURSE-LED SERVICES IN PSYCHIATRY counselling – family therapy counselling – cognitive behavioural therapy addiction service rehabilitation service attention deficit hyperactivity disorder service bereavement service

Framework for establishing nurse-led/midwife-led services

Introduction

Taking into account the findings of this study, national policy documents and the international literature the following is a recommended framework for organisations or services intending to introduce nurse-led and/or midwife-led services in the future.

New models of care require a review of the competencies and skills of nurses and, midwives and, where appropriate, expansion of practice. When identifying areas for expansion of practice, frameworks exist to support this:

- Scope of practice (ABA 2000a)
- Code of conduct (ABA 2000b)
- ANP/AMP framework (2nd ed) (National Council 2004a)
- CNS/CMS frameworks (2nd ed) (National Council 2004b)
- ABA guidelines for midwives (ABA 2001)

Underpinning these frameworks are evidence-based protocols and multidisciplinary integrated care planning. Examples of their use to support nurse-led/midwife-led care include areas such as medication management, referrals to the multidisciplinary team, requesting of tests and admission and discharge of the patient/client.

It is recommended that a business plan approach is adopted to aid the development of nurse-led/midwife-led services. The following provides a template to assist this. Organisations should utilise this in conjunction with their own internal processes.

Preparing a business case for the introduction of nurse-led/midwife-led services

In developing the business case the following four principles should be adhered to:

- 1. The outcome for the patient/client is the top priority.
- 2. The service to be provided will be demonstrably cost-effective.
- **3.** The service model or care delivery model will be based on evidence relating to the needs of the specific population and/or the caseload.
- 4. Models of evidence-based best practice will be adopted.

A project management approach to develop the business case should be utilised. This entails identifying stakeholders, establishing a project team and setting targets to be achieved within a stated and agreed timeframe. Members of the project team should be chosen according to their particular area of expertise, ensuring that a multidisciplinary and/or interdisciplinary approach is adopted. Finance and human resources departments can provide expertise, as they have the required knowledge. A total systems approach will achieve results more successfully and effectively.

The content of the business case should include:

- Service needs assessment
- Human resource implications, including an analysis of the skills and competencies that will be required
- Financial analysis
- Non-financial analysis
- Evidence and risk
- Implementation plan

The following section provides a template for a business case. If the organisation has its own template, this should be used. There is no single 'right' outline, format or content list when writing a business plan. The content needs to be credible, accurate, logical and succinct.

Template for a business case

Proposition or summary

A statement of the new service that is being proposed should be written at the early stages. This should briefly detail the new service and anticipated benefits to the service. The statement should be written with the target audience in mind.

Context

Include a brief statement about why the proposed change really matters to patient/client care and the organisation. The geographic location affected by the proposition (or not) should be addressed and the organisational context outlined.

Service needs analysis

This section needs to demonstrate the need for the new service. How the new service will impact on patient/client care should be outlined. Provide details of current service provision, current activities, client/patient need and gaps in service in order to demonstrate what the impact on patients/clients will be. This information could come from the following sources:

Service data

- current activity levels
- current waiting list figures
- results of audits
- caseload analysis
- population demographics particular to the service
- epidemiological data available and particular to the service, including mortality and morbidity rates at local, regional, national and international levels.

Information

- Hospital In-Patient Enquiry (HIPE)
- National Cancer Registry
- Public health information systems
- Central Statistics Office
- Local health statistics

Health and social policy documents and reports⁷

- Evaluation of the Irish Pilot Programme for Health Care Assistants (2004)
- Health Strategy Quality & Fairness A Health System For You (2001)
- Cardiovascular Strategy (1999)
- Cancer Strategy (1996)
- Primary Care Strategy (2001)
- Palliative Care Strategy (2001)
- AIDS Strategy (2000)
- National Breastfeeding Policy (1994)
- National Health Information Strategy (2004)
- The National Health and Life Styles Survey (2003)
- Research Strategy for Nursing and Midwifery in Ireland (2003)
- Acute Hospital Bed Capacity A National Review (2003)
- Health Statistics (2002)
- Long Stay Activity Report (2001)
- Effective Utilisation of the Professional Skills of Nurses and Midwives (2001)
- European Home and Leisure Accident Surveillance System Report for Ireland (2002)

Include a statement of service overview and activity analysis, indicating the level of service currently provided, followed by the projected level of service that will be provided by the introduction of the new service.

Financial analysis

This is a critical section of the business case. Accurate data must be provided in relation to the scale of new service, i.e. new posts, resources, etc. Utilise the assistance of the finance department in the organisation.

Estimated costs split between:

• Non-recurring (one-off) costs: project management, equipment, recruitment, initial training and evaluation, changes to accommodation, 'pump priming' and continuing costs, salaries etc.

Estimated savings:

• Estimated savings can be more difficult to identify than costs. Identify ways of doing things differently. Look at what the organisation is currently spending which is often very different to what is budgeted and what could be saved over time.

Look for the savings in staff costs such as:

• Reduced use of agency and locum staff, reduced staff turnover and from reducing multiple visits by the patients/clients to hospitals, less complaints, less paperwork.

Timing:

• An analysis of costs and savings over the relevant financial years. If unsure, make an estimate.

	2004-5	2005-6	2006-7
Non-recurring costs			
Continuing costs			
Savings			

Non-financial impact

• Quantify the likely impact of the change on key performance areas such as quality, reduced waiting times, increased patient/client satisfaction and clinical performance indicators.

Human resources implications

The Health Strategy recommends having the right person with the right level of knowledge and the right expertise to lead the service from a clinical perspective. In many cases a business proposal will be about re-engineering the way in which people work and maximising the potential available competencies. To this end a good business case should include the competencies and skills that will be required to deliver the proposed service. This should include the level of decision making expected and the level of autonomy.

The availability of education to ensure the required competencies and skills is a necessity. The business case will therefore need to identify educational needs within the organisation. These may be delivered 'in-house', by the centres for nurse education, third level institutions or other means. Agreements with the education providers should be outlined along with the cost implications associated with fees, replacement costs and time.

Evidence and risk

• Detail how the proposed change will work. Give examples of small-scale tests or history of success elsewhere. Also include potential risks and contingency plans to prevent them.

Quality improvement information

- Provide evidence from the international literature demonstrating the efficacy (or otherwise) of similar services
- Outline patient/client expectation of the service
- Outline the perceived contribution of the new service to patient/client care
- Discuss what the introduction of a new service will bring to the organisation that was not already there
- Review and critically compare other similar services within the region or nationally
- Discuss how the introduction of the service will be monitored. It is important to collect baseline data for comparative purposes

Implementation plan

- Outline the timeframe for delivery of the new service from approval of business plan to initiation of new service. Use Gantt charts as appropriate.
- Discuss the business plan with key decision makers in your organisation prior to finalising it.
- Submit the business plan to key decision makers as appropriate.
- Make recommendations for inclusion in the regional or local service plan if appropriate.

Summary

Most business plans require review and amendments as they go through the formal review processes by key decision makers and stakeholders. Be prepared for this and be open to new suggestions. It takes time to prepare, submit and get approval for a business plan – remain committed to the justifications for the new service and clearly articulate the vision which will improve patient/client care. Utilise all opportunities with key stakeholders and decision makers to promote the business plan.

Many factors such as the ageing population, population increases, changing ethnic composition and increasing urbanisation will affect the requirement for nursing and midwifery services and specialisation within health services in the future. This framework has indicated the types of information and sources of evidence that will be required when a new service is being proposed. The HSRP and other national and regional strategies and policies, regional demographic and epidemiological profiles and local requirements based on service need are the foundation of any proposal.

All new services must have an evaluation strategy built in to assist the process of continuous quality improvement and to ensure that the service fits the patient/client need. This evaluation should be as holistic as possible, incorporating the views of patients/clients and communities. The introduction of any new service will have implications for the wider system of working and cannot be considered in isolation.
References

An Bord Altranais (2000a) The Scope of Nursing and Midwifery Practice Framework. An Bord Altranais, Dublin

An Bord Altranais (2000b) The Code of Professional Conduct for Each Nurse and Midwife. An Bord Altranais, Dublin

An Bord Altranais (2001) Guidelines for Midwives 3rd ed., An Bord Altranais, Dublin

Baldwin R., Pratt H., Marriott A. Roberts C. (2001) Does a nurse-led mental health liaison service for older people reduce psychiatric morbidity in acute general medical wards. A randomised controlled trial. *Age and Ageing* 33: 472-478.

Biro A., M., Waldenstrom U., Brown S., Pannifex J. (2003) Satisfaction with Team Midwifery Care for Low -and High- Risk women: A Randomised Controlled Trial. *Birth* March 1-9

Biro M., A., Waldenstrom U., Pannifex J. (2000) Team Midwifery Care in a Tertiary Level Obstetric Service: A Randomized Controlled Trial. *Birth* 27 (3) 168-173

Blue, L., Lang., E., Mc Murrary, J.J., et al (2001) Randomised controlled trial of specialist nurse intervention in heart failure. *British Medical Journal* Sep 29; 323:715-8

Bostrom, B., Lundberg, D., Fridlund, B. (2003) Pain and health-related quality of life among cancer patients in final stage of life: a comparison between two palliative care teams. *Journal of Nursing Management*. Vol. 11 (3) 189-196

Boyer H., and Whiles L. (2004) Nurse-led assessment of children receiving chemotherapy. *Paediatric Nursing* June 16:5 26-27

Breen, A., Carr, E., Mann, E., Crossen, H. (2004) Acute back pain management in primary care: a qualitative pilot study of the feasibility of a nurse-led service in general practice. *Journal of Nursing Management*. Vol. 12 (3) 201-209

Byrne J., P., Crowther C., A., Moss J., R. (2000) A randomised controlled trial comparing birthing centre care with delivery suite care in Adelaide, Australia. *Australian and New Zealand Journal of Obstetrics and Gynaecology* 40 (3) 268-74

Campbell R., Macfarlane A., Hempsall V., Hatchard K. (1999) Evaluation of midwife-led care provided at the Royal Bournemouth Hospital. *Midwifery* 15 (3) 183-93

Campbell, N., C., Thain, J., Deans, H., G., et al. (1999) Secondary Prevention clinics for coronary heart disease: randomized trial of effect on health . *BMJ* 1998 May 9;316:1434-7 Accessed 06/07/2004 http://ebn.bmjjournals.com *Evidence-Based Nursing* 1999; 2:21

Clarke T., Baker P., Watts C., J., and Williams K. (2002) Self-harm in adults: A randomised controlled trial of nurseled case management versus routine care only. *Journal of Mental Health* 11,2: 167

Clegg A., and Mansfield S. (2003) Nurse consultants and the care of older people: developing new roles. *Nursing Older People* June vol 15 no 4 14-16

Comhairle Na nOspideal (2003) Report of the Committee Reviewing Maternity and Related Services in the North Eastern Health Board Area

Coulthard M., G., Vernon S., J., Lambert H.J., Matthews J., N., S., *British Medical Journal.* (2003). Vol. 327, Iss. 7416; 656-9

Cox, K., Wilson, E. (2003) follow-up for people with cancer: nurse-led services and telephone interventions. *Journal of Advanced Nursing*. Vol. 43 (1) 51

Currie, M., P., Karwatowski, S., P., Perea, J., Langford, E.,J. (2003) Nurse-led DC cardioversion in a day treatment centre: a study of the safety and efficacy. *Heart.* Vol.89

Davies-Gray M (2003). Nurse led fast track for vulnerable older people. Emergency Nurse. Vol. 11, Iss. 5; 34-38

Delamere, S. (2004) Facing the Challenge of Change: Acceptability of the role of the Advanced Nurse Practitioner (Sexual Health) Poster presentation. The National Council for the Professional Development of Nursing and Midwifery 4th Annual Conference, Dublin

Denver, E., A., Barnard, M., Woolfson, R., G., Earle, K. A. (2003) Management of uncontrolled hypertension in a nurse-led clinic compared with conventional care for patients with type 2 diabetes. *Diabetes Care*. Vol.26, Iss. 8

Department of Health and Children (1994) A National Breastfeeding Policy for Ireland Stationery Office, Dublin

Department of Health and Children (1996) An Evaluation of Cancer Services in Ireland: a National Strategy Stationery Office, Dublin

Department of Health and Children (1999) Cardiovascular strategy Building Healthier Hearts Stationery Office, Dublin

Department of Health and Children (2000) AIDS Strategy Report of the National AIDS Strategy Committee Stationery Office, Dublin

Department of Health and Children (2001a), Quality and Fairness: A Health System for You Stationery Office, Dublin

Department of Health and Children (2001b) Primary Care: A New Direction Stationery Office, Dublin

Department of Health and Children (2001c) Long Stay Activity Report Stationery Office, Dublin

Department of Health and Children (2001d) Report of the National Advisory Committee on Palliative care Stationery Office, Dublin

Department of Health and Children (2001e) *Effective Utilisation of the Professional Skills of Nurses and Midwives* Stationery Office, Dublin

Department of Health and Children (2002a) Health Statistics Stationery Office, Dublin

Department of Health and Children (2002b) European Home and Leisure Accident Surveillance System Report for Ireland Stationery Office, Dublin

Department of Health and Children (2003a), Report of the National Task Force on Medical Staffing Stationery Office, Dublin

Department of Health and Children (2003b), The Challenge for Nursing and Midwifery A Discussion Paper (NP)

Department of Health and Children (2003c), Empowerment Narratives. A collection of exemplars illustrating empowerment of nurses and midwives Stationery Office, Dublin

Department of Health and Children (2003d) The National Health and Life Styles Survey Stationery Office, Dublin

Department of Health and Children (2003e) Research Strategy for Nursing and Midwifery in Ireland Stationery Office, Dublin

Department of Health and Children (2003f) Acute Hospital Bed Capacity - A National Review Stationery Office, Dublin

Department of Health and Children (2004a) *Evaluation of the Irish Pilot Programme for Health Care Assistants* Stationery Office, Dublin

Department of Health and Children (2004b) National Health Information Strategy Stationery Office, Dublin

Department of Health (1994) Report of the Mother and Infant Care Scheme Review Group. Dublin: Department of Health, Stationery Office, Dublin

Department of Health (1993) Changing Childbirth Final Report of the Expert Maternity Group London: Her Majesty's Stationery Office

Dobson, R. (1999) Patients satisfied with nurse run practices British Medical Journal. Vol.319, Iss. 7212

Fitzmaurice, D., A., Hobbs, F., D., Murray, E., T. Oral anti-coagulation management in primary care with the use of computerized decision support and near patient testing: a randomized controlled trial. Arch Intern Med (2000) Aug 14-28; 169: 2343-8 Accessed 06/07/2004 Evidence-Based Nursing http://ebn.bmjjournals.com *Evidence-Based Nursing* 2003 4:57

Forster, A., Young, J. (1996) Specialist nurse support for patients with stroke in the community: a randomized controlled trial. *British Medical Association* 312, 1642-1646

Gallagher A. (1999) Nurse-led care for children with asthma. Paediatric Nursing; 11:8 Oct 36-37

Garvican, L., Grimby, E., Littlejohns, P., Lowndes, S., Sachs, N. (1998) Satisfaction with clinical nurse specialists in a breast care clinic: questionnaire survey. *British Medical Journal* 316: 7136, 976-977

Government of Ireland (1998), *Report of The Commission on Nursing 1998 – A blueprint for the future*, Stationery Office, Dublin

Griffin G. (2001) Introducing a nurse-led deliberate sel-harm assessment service. Psychiatric Bulletin 25:212-214.

Griffiths, H.(2002) Iron deficiency anaemia: developing a nurse-led integrated care pathway. *Nursing Standard*. Vol. 16, Iss, 45 38

Griffiths, P., Harris, R., Richardson, G., Hallet, N., Heard, S., Wilson-Barnett, J. (2001) Substitution of a nursing-led inpatient unit for acute services: randomised controlled trial of outcomes and cost of nursing-led intermediate care. *Age and Ageing* 30 (6) November

Hall-Smith, J., Ball, C. and Coakely, J. (1997) Follow up services and the development of a clinical nurse specialist in intensive care. *Intensive and Critical Care Nursing* 13, 243-248

Harrison, M., B., Browne, G., B. (2002) Quality of life of individuals with heart failure: a randomised trial of the effectiveness of two models of hospital-to-home transition. Accessed 06/07/2004 http://ebn.bmjjournals.com *Evidence-Based Nursing* 2003; 6:21

Harvey S., Jarrell J., Rollin B., Stainton C., Rach D. (1996) A Randomised, Controlled Trial of Nurse-Midwifery Care. *Birth* 23 (3) 128-135

Hatem M., Hodnett E., D., Devane D., Fraser W., D., Sandal J., Soltani H. (2004) Midwifery-led versus other models of care delivery for childbearing women. *The Cochrane Database of Systematic Reviews* 2004, Issue 1

Hodnett E., D. (2001) Continuity of care givers for care during pregnancy and childbirth (Cochrane Review) In: The Cochrane Library, Issue 3 2001 Oxford: Update Software CD000062

Homer C., S., Davis G., K., Cooke M., Barclay M., B. (2002) Women's experiences of continuity of midwifery care in a randomised controlled trial in Australia. *Midwifery* 18 102-112

Horrocks, S., Anderson, E., Salisbury, C (2002) systematic review of whether nurse practitioners working in primary care can provide equivalent care to doctors. *British Medical Journal* 324:819-823 (6 April)

House of Commons Select Committee on Health (1992) Second Report on the Maternity Services (Winterton report) London: Her Majesty's Stationery Office

Hundley V., A., Cruickshank F., M., Lang G., D., CMA, Glazner CMA Milne J., M., Turner M., Blyth D., Mollison J., Donaldson C. (1994) Midwife managed delivery unit: a randomised controlled comparison with consultant led care. *British Medical Journal* 309: 1400--1404

Hundley V., A., Cruickshank F., M., Milne J., M., et al (1995) Satisfaction and continuity of care: staff views of care in a midwife managed delivery unit. *Midwifery* 11: 163-173

Kerr, M., Fraser, W., I., Felce, D. (1996) Primary Health Care for People with Learning Disabilities. *British Journal of Learning Disabilities*, 24, 2-8

Kenealy, T., Kenealy, H., Docherty, B., Scott, D., Scragg, R., Simmons, D. (2004) Diabetes care: practice nurse roles, attitudes and concerns. *Journal of Advanced Nursing*. Vol. 48 (1) 68-75

Leveille, S., G., Wagner, E., H., Davies, C., et al (1998) preventing disability and managing chronic illness in frail older adults: a randomised trial of a community-based partnership with primary care. *Journal American Geriatric Society* Oct; 46:1191-8eilleeille

Mac Lellan, K. (2004) Postoperative pain: Strategy for improving patient experiences. *Journal of Advaced Nursing*. Vol 46 (2) 179-185

McHugh, F., Lindsay, G., M., Hanlon. P., Hutton, I. et al (2001) Nurse-led shared care for patients on the waiting list for coronary artery bypass surgery: a randomised controlled trial. *Heart.* Vol.86, Iss.3 317-324

Mackintosh, C., Bowles, S. (1997) Evaluation of a nurse-led acute pain service. Can clinical nurse specialists make a difference? *Journal of Advanced Nursing* Vol. 25 (1) 30-37

Madge, P., McColl, J., Paton, J. (1998) Impact of a nurse-led home management training programme in children admitted to hospital with acute asthma: a randomised trial. Accessed 06/07/2004 http://ebn.bmjjournals.com *Evidence-Based Nursing* 1998;1:11

Marshall, D., Moore, G., McConkey, R. (2003a) Integration and the Healthcheck 2000 Experience: Breaking through to Better Health for People with Learning Disabilities, in Association with Primary Care. In Brown, M (Ed) *Learning Disability: A Handbook for Integrated Care*, APS Publishing, Salisbury

Marshall, D., McConkey, R., Moore, G. (2003b) Obesity in People with Intellectual Disabilities: The Impact of Nurse-Led Health Screenings and Health Promotion Activities. *Journal of Advanced Nursing*, Vol 41 (2), 147-153

Martin, G, Phillip, L, Bates, L, Warwick, J (2004) Evaluation of a Nurse Led Annual Review of Patients with Severe Intellectual Disabilities, Needs Identified and Needs Met, in Large Group Practice. *Journal of Learning Disabilities* 8(3) 235-246

Mencap (2004) Treat Me Right! Better Healthcare for People with a Learning Disability, Mencap, London

Miles, K., Penny, N., Power, R., Mercey, D. (2003) Comparing doctor and nurse-led care in a sexual health clinic: patient satisfaction questionnaire. *Journal of Advanced Nursing*. Vol. 42 (1) 64-72

Mills, N., Campbell, R., Bachmann, M. (2002) Professional and organisational obstacles to establishing a new specialist service in primary care: case study of an epilepsy specialist nurse. *Journal of Advanced Nursing*. Vol. 37 (1) pp.43-51

Moore, S., Corner, J., Haviland, J., Wells, M. et al (2002) Nurse led follow up and conventional medical follow up in management of patients with lung cancer: randomised trial. *British Medical Journal*. Vol. 325, Iss, 7373 1145

Morgan M., Fenwick N., McKenzie C., Wolfe C., D. (1998) Quality of midwifery led care: assessing the effects of different models of continuity for women's satisfaction. *Quality in Health Care* 7 (2) 77-82

MWHB (2003) An Explorative Study into the Expansion of Nursing and Midwifery Professional Roles in Response to the European Working Time Directive. MWHB

National Council for the Professional Development of Nursing and Midwifery (2004a), *Framework for the Establishment of Advanced Nurse/Midwife Practioner Posts* (2nd ed), National Council for the Professional Development of Nursing and Midwifery, Dublin

National Council for the Professional Development of Nursing and Midwifery (2004b), *Framework for the Establishment of the Clinical Nurse/Midwife Specialist* (2nd ed), National Council for the Professional Development of Nursing and Midwifery, Dublin

National Council for the Professional Development of Nursing and Midwifery (2003), Agenda for the Future Professional Development of Nursing and Midwifery, National Council for the Professional Development of Nursing and Midwifery, Dublin

National Disability Authority (2004) Towards Best Practice in Provision of Health Services for People with Disabilities in Ireland, National Disability Authority, Dublin

National Institute for Mental Health in England (2005) Nurse-led mental health assessment clinic, Royal United Hospital, Bath. Cited on the world wide web http://www.nimhe.org.uk/smartsearch/itemdisplay_story.asp?id=116

O'Connell R., Cronin M. (2002) Home birth in Ireland 1993-1997: a review of community midwifery practice. *The All Ireland Journal of Nursing and Midwifery* 2 (2) 41-46

Pritchard, A., Kendrick, D. (2001) Practice nurse and health visitor management of acute minor illness in a general practice. *Journal of Advanced Nursing*. Vol. 36 (4) 556-562

Rando, T. (2000) A healing success- RDNS would assessment clinic. Australian Nursing Journal Vol. 7 Iss. 10 33

RCOG (2001)Hatem M., Devane D., Fraser W., D., Hodnett E., D., Sandall J., Soltani H. Midwifery-led versus other models of care delivery for childbearing women (Protocol for a Cochrane Review). In: *The Cochrane Library*, Issue 1, 2004. Chichester, UK: John Wiley and Sons, Ltd.

RCOG (2001) Royal College of Obstetricians and Gynaecologists. The National Sentinel Caesarean Section Audit Report. RCOG Clinical Effectiveness Support Unit 2001

Richardson, A., Cunliffe, L. (2003) New Horizons: the motives, diversity and future of nurse-led care. *Journal of Nursing Management* Vol. 11 (2) March

Roxburgh, M., (2000) Developing a nurse-led cystoscopy service *British Journal of Perioperative Nursing* Vol. 10, Iss 5 246-53

Saunders D., Boulton M., Chapple J., Radcliffe J., Levitan J. (2000) Evaluation of the Edgware birth Centre Middlesex: North Thames Perinatal Public Health

Shields, N., Turnbull, D., Reid, M., Holmes, A., McGinley M., Smith L., N. (1998) Satisfaction with midwifemanaged care in different periods: a randomized controlled trial of 1299 women. *Midwifery* 14, 85-93

Smith, G., Watson, R., Roger, D., McRorie, E., Hurst, N., Luman, W., Palmer, K. (2002) Impact of a nurse-led counselling service on quality of life in patients with inflammatory bowel disease. *Journal of Advanced Nursing* Vol. 38 (2) pp.152-160

Sutcliffe, A., M. (1999) A regional nurse-led osteoporosis clinic. Nursing Standard. Vol. 13 Iss.37

The Northern and Yorkshire Regional NHS Modernisation Programme (2005) Nurse-led sexual health assessment to improve compliance with psychotrophic medication. Cited on the world wide web http://www.nyx.org.uk/modernprogrammes/mentalhealth/goodpractice/may2002/nurseled.html

Todd C., J., Farquher M., C., Camilleri-Ferrante C. (1998) Team midwifery: the views and job satisfaction of midwives. *Midwifery* 14 (4) 214-24

Tummey R. (2001) A collaborative approach to urgent mental health referrals. Nursing Standard 15(52) 39-42.

Turnbull D., Holmes A., Shields N., Cheyne H., Twaddle S., Harper-Gilmour W., McGinley M., Reid M., Johnstone I., Geer I., McIlwaine G., Burnett Lunan C. (1996) Randomised, controlled trial of efficacy of midwife-managed care *The Lancet* 348 213-218

Twinn, S., Cheng, F. (1999) A case study of the effectiveness of nurse-led screening programmes for cervical cancer among Hong Kong Chinese women. *Journal of Advanced Nursing*. Vol.29 (5) 1089-1096

Van Schrojenstein Lanteman-de Valk, H., Metsemakers, J., F., Haveman, M., J., Crebolder, H., F. (2000) Health Problems in People with Intellectual Disability in General Practice: A Comparative Study, *Family Practice*, 17(5), 405-407

Villar J., Carroll G., Khan-Neelofur D., Piaggio G., Gulmexoglu M. (2004) Patterns of routine antenatal care for lowrisk pregnancy (Cochrane Review) In: *The Cochrane Library* Issue 2, 2004 Chichester, UK: John Wiley & Sons Ltd.

Waldenstrom U., Brown S., Mclachlan J., Forster D., Brennecke S. (2000) Does Team Midwife Care Increase Satisfaction with Antenatal, Intrapartum and. Postpartum Care? A Randomised Controlled Trial. *Birth* 27:3 156-167

Waldenstrom U., Turnbull D. (1998) A systematic review comparing continuity of midwifery care with standard maternity services. *British Journal of Obstetrics and Gynaecology* Vol. 105: 1160-1170

Waldenstrom, U., Nilsson, C., Winbladh, B. (1997) The Stockholm Birth Centre Trial: maternal and infant outcome. *British Journal of Obstetrics and Gynaecology* Vol. 104 410-418 Walker J. (2001) Edgware Birth Centre: what is the significance of this model? Midwifery Digest 11 (1) 8-12

Walker J., Hall S., Thomas A. (1995) The experience of labour: a perspective from those receiving care in a midwife-led unit *Midwifery* 1 (1)120-129

Walsh, P., N., Emerson, E., Robertson, J. (2004) A Good Place to Live: The Quality of Residential Supports. In Walsh, P N, & Gash, H (Eds) *Lives and Times: Practice, Policy and People with Disabilities*, Rathdown Press, Bray, 276-295

Walsh, P., N., Linehan, C., Hillery, J., et al (2000) Quality and Costs of Residential Supports for Irish Adults with Intellectual Disability. Brief Summary Report. Cited by Walsh, P N, Emerson, E, Robertson, J (2004) A Good Place to Live: The Quality of Residential Supports. In Walsh, P N, & Gash, H (Eds) *Lives and Times: Practice, Policy and People with Disabilities*, Rathdown Press, Bray, 276-295, 288

Watts K., Fraser D., M., Munir F (2003) The impact of the establishment of a midwife-managed unit on women in a rural setting in England *Midwifery* 19 (2) 106-112

WHO (1999) World Health Organisation Care in normal birth: a practical guide Geneva: *World Health Organisation* 1999

Wiles, R., Postle, K., Steiner, A., Walsh, B.(2003) Nurse-led intermediate care: patients perceptions. *International Journal of Nursing Studies* 40 61

Williams, R., Assassa, R., Smith, N., Rippin, C. (2002) Good practice in continence care: development of nurse-led service. *British Journal of Nursing*. Apr 25-May 8 Vol. 11, Iss. 8 548

Wilson-Barnett, J., Davies, T., Griffiths, P., Harris, H., Heard, S. (2001) Evaluating nurse-led inpatient units. *Nursing Standard* May 30- June 5, Vol. 15 iss.37; 40

Winterton report (1992) House of Commons Health Select Committee, Second Report on maternity Services, London: HMSO

York, R. Brown, L.P., Samuels, P., et al. (1998) A randomised trial of early discharge and nurse specialist transitional follow-up care of high-risk childbearing women. Nurs Res sep/oct;46:254-61 Accessed 06/07/2005 http://ebn.bmjjournals.com *Evidence-Based Nursing* 1998;1:73

Young, D., Shields, N., Holmes, A., Turnbull D., Twaddle S. (1997) A new style of midwife-managed antenatal care: costs and satisfaction. *British Journal of Midwifery* 5 (9) 540-545

AN EVALUATION OF THE EXTENT AND NATURE OF NURSE-LED/MIDWIFE-LED SERVICES IN IRELAND

APPENDICES Appendix 1: Nurse-led Questionnaire

National Council for the Professional Development of Nursing and Midwifery

Nurse-led Care questionnaire

	Respondent's Details No: NO.
	Prefix First Name
Address details	Surname
	Title
	Organisation
Dear A Another	
The National Council for the Professional Development of Nursing	
and Midwifery is undertaking a research study to establish the	Address
extent of nurse-led services in Ireland.	Address
Nurse-led care is distinct from nurse coordinated or nurse-managed	
services. Nurse-led care is provided by nurses responsible for case	
management, which included comprehensive patient/client	
assessment, developing, implementing and managing a plan of	
care, clinical leadership and decision to admit or discharge.	
Patients/clients will be referred to nurse-led services by nurses,	
midwives or other healthcare professionals, in accordance with	
collaboratively agreed protocols. Such care requires increased skills	If any of the above details need to be amended
and knowledge and the nurse will need preparation in both the	please fill out the fields below.
clinical and management aspects of the role. Such nurses will be	Prefix First Name
practicing at an advanced level and may be working in specialist or	
advanced practice roles. (National Council 2003)	Surname
To this end I would be very grateful if you or a nominated colleague	
would complete this questionnaire.	Title
If you have any questions concerning the content, please don't	
hesitate to contact me on 01 882 5314.	Organisation
Please return the questionnaire in the FREEPOST envelope provided.	
Yours sincerely	Address
Hora.	
Jenny Hogan	
Professional Development Officer	

Q1: In which of the f	ollowing care settings does your	organisation/service belong?	
Community Hos	pital 📃 Hospital & Community 🗌		
Other (please specify)			
Q2: To which of the	following bands does your organi	isation/service belong to?	
Band 1 Band 2	Band 3 Band 4	Band 5	
Other (please specify)			
Q3: How many beds	are there in your organisation/se	rvice?	
50 -100 🗌 101-150			
351-400 401-50	0 More than 500 Not	applicable	
Q4: Do nurses provid	le a nurse·led service? (see cove	ring letter for definition of nurs	e-led care)
Yes No	Don't know If you have ticked	No or Don't know please skip to que	stion 16
		No or boil i know please skip to que	5000 10
Q5: If you ticked yes	to question 4 please detail each	n service you consider to be nur	se-led and how
Q5: If you ticked yes	s to question 4 please detail each it been established?		
Q5: If you ticked yes many years has	to question 4 please detail each		se-led and how YEARS ESTABLISHED
Q5: If you ticked yes	s to question 4 please detail each it been established?		
Q5: If you ticked yes many years has Service 1	s to question 4 please detail each it been established?		
Q5: If you ticked yes many years has Service 1 Service 2 Service 3	s to question 4 please detail each it been established?		
Q5: If you ticked yes many years has Service 1 Service 2	s to question 4 please detail each it been established?		
Q5: If you ticked yes many years has Service 1 Service 2 Service 3 Service 4 Service 5	s to question 4 please detail each it been established?		
Q5: If you ticked yes many years has Service 1 Service 2 Service 3 Service 4	s to question 4 please detail each it been established?		
Q5: If you ticked yes many years has Service 1 Service 2 Service 3 Service 4 Service 5 Service 6	s to question 4 please detail each it been established?		YEARS ESTABLISHED
Q5: If you ticked yes many years has Service 1 Service 2 Service 3 Service 4 Service 5 Service 6	s to question 4 please detail each it been established? TITLE OF SERVICE		YEARS ESTABLISHED
Q5: If you ticked yes many years has Service 1 Service 2 Service 3 Service 4 Service 5 Service 6 Q6: Do the nurses w Service 1	to question 4 please detail each it been established? TITLE OF SERVICE	e following? (Please tick accordin COMPREHENSIVE ASSESSMENT	YEARS ESTABLISHED
Q5: If you ticked yes many years has Service 1 Service 2 Service 3 Service 4 Service 5 Service 6 Q6: Do the nurses w Service 1 Service 2	to question 4 please detail each it been established? TITLE OF SERVICE	e following? (Please tick accordin COMPREHENSIVE ASSESSMENT	YEARS ESTABLISHED
Q5: If you ticked yes many years has Service 1 Service 2 Service 3 Service 4 Service 5 Service 6 Q6: Do the nurses w Service 1 Service 2 Service 3	to question 4 please detail each it been established? TITLE OF SERVICE	e following? (Please tick accordin COMPREHENSIVE ASSESSMENT	YEARS ESTABLISHED
Q5: If you ticked yes many years has Service 1 Service 2 Service 3 Service 5 Service 6 Q6: Do the nurses w Service 1 Service 2 Service 3 Service 4 Service 5 Service 6 Service 1 Service 2 Service 3 Service 4	to question 4 please detail each it been established? TITLE OF SERVICE	e following? (Please tick accordin COMPREHENSIVE ASSESSMENT	YEARS ESTABLISHED
Q5: If you ticked yes many years has Service 1 Service 2 Service 3 Service 4 Service 5 Service 6 Q6: Do the nurses w Service 1 Service 2	to question 4 please detail each it been established? TITLE OF SERVICE	e following? (Please tick accordin COMPREHENSIVE ASSESSMENT	YEARS ESTABLISHED
Q5: If you ticked yes many years has Service 1 Service 2 Service 3 Service 5 Service 6 Q6: Do the nurses w Service 1 Service 2 Service 4 Service 5 Service 6 Service 7 Service 1 Service 2 Service 3 Service 4 Service 5	to question 4 please detail each it been established? TITLE OF SERVICE	e following? (Please tick accordin COMPREHENSIVE ASSESSMENT	YEARS ESTABLISHED

Q8: Do nurses who run nu						
Q8: Do nurses who run nu						
	rse-led	services discharge	their patient	ts/clie	ents?	
(Discharge can mean from any he	ealthcare	institution or from the e	episode of care t	hat the	e nurse was providing)	
Yes No Don't	know]				
Q9: Please tick to whom t		oo wuxning the nuw	no lod olinio d			
Dietitian		Community mental h	ealth nurse		Consultant	
Physiotherapist		Public health nurse			No-one	
CNS's/CMS's		Social worker			Other (please specify)	
Occupational therapist		Psychologist				
Speech therapist		Medical staff				
Radiographer		GP				
Q10: Please tick from who	om the r	ourse running the n	urse-led clini	ic car	receive referrals from:	
Dietitian		Community mental h	leaith nurse		CNM 1,2 or 3 Self referrals from	
Physiotherapist CNS's/CMS's					patients/clients/relatives	
		Psychologist GP			No-one	
Occupational therapist		Consultant			Other (please specify)	
Speech therapist Radiographer		Other Medical staff				
Public health nurse		Staff nurses				
Tublic fleatin huise		Stan huises				
Q11: Are the nurses who r	un nurs	e-led services wor	king as eithe	r:		
(You may tick more than one whe						
1. a staff nurse			4. a clinical nu	irse ma	anager	
2. a clinical nurse specialist			5. none of the			
3. an advanced nurse practitioner					ot listed above	
				, ,		
				ur org	anisation received education	or
training specifically	to equip	them to lead the	service?			
Yes No Don't	know]				
	10 010			otion	or training that the individual	
undertook?		could you list the r	elevant educ	ation	or training that the individual	
Certificate course			Post graduate	diplon	na	
In-house training			Masters degre	-		
Diploma			PhD			
Degree			None			

Nurse-led Care questionnaire
Q14: If you have nurse-led services are audits performed?
Yes No
Q15: Would you be willing to allow the National Council access to the audit reports or summaries of reports?
Yes No N/A
Q16: Is your organisation developing any nurse-led services?
Yes No Don't know
Q17: If you ticked yes to the previous question could you give a brief outline of the services under development?
Q18: Does your organisation/service have a patient/client advocacy group?
Yes No Don't know
Q19: If you answered yes to Q18 would you be willing to discuss this at a later date with the National Council regarding access to patient/client information?
Yes No N/A
Q20: Phase two of the research will consist of focus groups with organisations that are providing nurse- led care. If you or your service providers would be interested in taking part in these focus groups please tick the relevant box.
Willing to take part in focus groups
NOT willing to take part in focus groups
National Council for the All information in this questionnaire will be treated confidentially.
Professional Development of Nursing and Midwifery 6/7 Manor Street Business Park, Manor Street, Dublin 7. An Chomhairle Naisiunta d'Ehorbairt Chairmiúil an Altranais agus
an Chnaimhseachais Email: conference@ncnm.ie Website: www.ncnm.ie

APPENDICES Appendix 2: Midwifery-led Questionnaire

National Council for the Professional Development of Nursing and Midwifery

Midwifery-led Care QUESTIONNAIRE

	Respondent's Details No: M0008
	Prefix First Name
Address Details	Surname
	Title
	Organisation
ear A Another	
ne National Council for the Professional Development of Nursing nd Midwifery is undertaking a research study to establish the stent of midwifery-led services in Ireland.	Address
idwifery-led care has been defined by the recent Cochrane rotocol as "the context of care where 'the midwife is the lead rofessional in the planning, organisation and control of the care ven to a woman from initial booking to the postnatal period' RCOG 2001). Within these models, midwives are, in partnership ith the woman, the lead professional with responsibility for	
esessment of her needs, planning her care with her, referral to her health professional as appropriate, and for ensuring provision maternity services".1	If any of the above details need to be amended please fill out the fields below. Prefix First Name
this end I would be very grateful if you or a nominated colleague ould complete this questionnaire.	Surname
you have any questions concerning the content, please don't esitate to contact me on 01 882 5314.	Title
lease return the questionnaire in the FREEPOST nvelope provided.	Organisation
burs sincerely	
anny Hogan rofessional Development Officer	Address
Hatem M, Devane D, Fraser WD, Hodnett ED, Sandall J, Soltani H. Midwifery-led versus other models of care delivery for childbearing women (Protocol for a Cochrane Review). In: The Cochrane Library, Issue 1, 2004. Chichester, UK: John Wiley and Sons, Ltd.	

Midwifery	-led C	are QUE	STIONNAIRE
		\sim	

Q2: To which of the f	ollowing bands does your orga	nisation/service belong to?	
Band 1 📃 🛛 Band 2 🗍	Band 3 Band 4	Band 5	
Other (please specify)			
		a muia a 2	
	are there in your organisation/s		
50 -100 101-150 351-400 401-500		lot applicable	
351-400 401-300			
Q4: Do midwives pro	vide a midwifery-led service? (see covering letter for definition of	midwifery-led care)
Yes No	Don't know If you have ticke	d No or Don't know please skip to que	stion 16
	to question 4 please detail ea it been established?	ch service you consider to be mid	wife-led and how
	TITLE OF SERVI	ICE	YEARS ESTABLISHED
Service 1			
Service 2			
Service 3			
Demilee 4			
Service 4			
Service 4 Service 5 Service 6			
Service 5			
Service 5	who run midwife-led services hav	ve the following? (Please tick accordin	ng to service as above)
Service 5	TOTAL RESPONSIBILITY	COMPREHENSIVE ASSESSMENT	PROVIDE CLINICAL
Service 5 Service 6 Q6: Do the midwives v			
Service 5 Service 6 Q6: Do the midwives v Service 1	TOTAL RESPONSIBILITY	COMPREHENSIVE ASSESSMENT	PROVIDE CLINICAL
Service 5 Service 6 Q6: Do the midwives v Service 1 Service 2	TOTAL RESPONSIBILITY	COMPREHENSIVE ASSESSMENT	PROVIDE CLINICAL
Service 5	TOTAL RESPONSIBILITY	COMPREHENSIVE ASSESSMENT	PROVIDE CLINICAL
Service 5 Service 6 Q6: Do the midwives v Service 1 Service 2 Service 3	TOTAL RESPONSIBILITY	COMPREHENSIVE ASSESSMENT	PROVIDE CLINICAL
Service 5 Service 6 Service 1 Service 2 Service 3 Service 4	TOTAL RESPONSIBILITY	COMPREHENSIVE ASSESSMENT	PROVIDE CLINICAL
Service 5 Service 6 Service 1 Service 2 Service 3 Service 4 Service 5 Service 6	TOTAL RESPONSIBILITY	COMPREHENSIVE ASSESSMENT PATIENT/CLIENT	PROVIDE CLINICAL

Q8: Do midwives who re	un midwife	ery-led services dis	charge their	patie	nts/clients?	
(Discharge can mean from any	/ healthcare	institution or from the e	pisode of care	that the	e midwife was providing)	
Yes No Do	on't know]				
Q9: Please tick to whom	n the mid	vives running the n	nidwifery-led	clinio	c can refer to:	
Dietitian		Community mental h	ealth nurse		Consultant	
Physiotherapist		Public health nurse			No-one	
CNS's/CMS's		Social worker			Other (please specify)	
Occupational therapist		Psychologist				
Speech therapist		Medical staff				
Radiographer		GP				
Q10: Please tick from w	/hom the I	nidwive running th	e midwifery-l	ed cli	inic can receive referrals fro	om;
Dietitian		Community mental h	ealth nurse		CMM 1,2 or 3	
Physiotherapist		Social worker			Self referrals from	
CNS's/CMS's		Psychologist			patients/clients/relatives	
Occupational therapist		GP			No-one	
Speech therapist		Consultant			Other (please specify)	
Radiographer		Other Medical staff				
Public health nurse		Staff midwives				
Q11: Are the midwives					ner:	
(You may tick more than one v	where the se	rvice is provided by mo				_
1. a staff midwife			4. a clinical m			
2. a clinical midwife specialist			5. none of the			
3. an advanced midwife practit	ioner		6. please spe	cify if n	ot listed above	
019 Hove the midwive				o in v	rouge organization received	
		cally to equip then			our organisation received ce?	
Yes No Do	on't know]				
		-				
Q13: If you answered ye undertook?	es to Q12	could you list the r	elevant educ	ation	or training that the individu	al
Certificate course			Post graduate	diplon	22	
In-house training			Masters degre		lia	
III HOUSE HAILING						
Diploma			PhD			

Midwifery-le	ed Care questionnaire
Q14: If you have midwifery.led	services are audits performed?
Yes No	
Q15: Would you be willing to al reports?	low the National Council access to the audit reports or summaries of
Yes No N/A	
Q16: Is your organisation devel	loping any midwifery-led services?
Yes No Don't know	
Q17: If you ticked yes to the pr development?	revious question could you give a brief outline of the services under
Q18: Does your organisation/se	ervice have a patient/client advocacy group?
Yes No Don't know	
	8 would you be willing to discuss this at a later date with the National to patient/client information?
Yes No N/A	
	h will consist of focus groups with organisations that are providing or your service providers would be interested in taking part in these the relevant box.
Willing to take part in focus groups NOT willing to take part in focus groups	
	I information in this questionnaire will be treated confidentially. any thanks for taking the time to fill out this questionnaire.
of Nursing and Midwifery – 6/ An Chomhairle Naisiúnta d'Fhorbairt Chairmiúil an Altranais agus	7 Manor Street Business Park, Manor Street, Dublin 7. elephone: 01 882 5300. Fax: 01 868 0366. mail: conference@ncnm.ie Website: www.ncnm.ie





National Council for the Professional Development of Nursing and Midwifery

An Chomhairle Náisiúnta d'Fhorbairt Ghairmiúil an Altranais agus an Chnáimhseachais